

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Campo, Francesco; Giunti, Sara; Mendola, Mariapia

Working Paper The Refugee Crisis and Right-Wing Populism: Evidence from the Italian Dispersal Policy

IZA Discussion Papers, No. 14084

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Campo, Francesco; Giunti, Sara; Mendola, Mariapia (2021) : The Refugee Crisis and Right-Wing Populism: Evidence from the Italian Dispersal Policy, IZA Discussion Papers, No. 14084, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/232836

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 14084

The Refugee Crisis and Right-Wing Populism: Evidence from the Italian Dispersal Policy

Francesco Campo Sara Giunti Mariapia Mendola

JANUARY 2021



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 14084

The Refugee Crisis and Right-Wing Populism: Evidence from the Italian Dispersal Policy

Francesco Campo

University of Milano Bicocca

Sara Giunti University of Milano Bicocca

Mariapia Mendola University of Milano Bicocca, IZA and CefES

JANUARY 2021

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

The Refugee Crisis and Right-Wing Populism: Evidence from the Italian Dispersal Policy^{*}

This paper examines how the 2014-2017 'refugee crisis' in Italy affected voting behaviour and the rise of right–wing populism in national Parliamentary elections. We collect unique administrative data throughout the crisis and leverage exogenous variation in refugee resettlement across Italian municipalities induced by the Dispersal Policy. We find a positive and significant effect of the share of asylum seekers on support for radical-right antiimmigration parties. The effect is heterogeneous across municipality characteristics, yet robust to dispersal policy features. We provide causal evidence that the anti–immigration backlash is not rooted in adverse economic effects, while it is triggered by radical–right propaganda.

JEL Classification:	D72, F22, O15, P16
Keywords:	immigration, refugee crisis, voting behavior, dispersal policy,
	impact evaluation

Corresponding author:

Mariapia Mendola Department of Economics Università di Milano Bicocca Piazza dell'Ateneo Nuovo 1 20126 Milano Italy E-mail: mariapia.mendola@unimib.it

^{*} We thank online seminar participants at the Collegio Carlo Alberto, University of Pavia, Insubria, Milano-Bicocca, ITSC–Ancona and GDS17–Rome for thoughtful comments and suggestions. We are grateful to openpolis, ActionAid (Italy) and Amnesty International Italy for their collaboration with this research. We also thank Federico Maggio, Navid Nobani, Elham Talebbeydokhti, Gianluca Tarasconi and Alessandro Vaccarino for invaluable support with data preparation. We acknowledge financial support from the World Bank and CefES. The usual disclaimer applies.

1 Introduction

During the 'refugee crisis' of 2014-2017, Europe faced an unprecedented influx of refugees and migrants. Around 3.5 million people applied for asylum in the EU-28 countries (Eurostat, 2020), most of them fleeing war and terror in Syria and social unrest in regions of North Africa and the Near East. This human inflow stretched the systems that were designed to manage asylum seekers, prompted a stream of reforms to strengthen existing reception structures and divided public opinion in many destination countries (Hatton, 2020; Dustmann et al., 2016).¹

While asylum applications peaked in 2016 and have fallen since,³ asylum seekers' arrivals decreased very unevenly among EU member states, with persistent pressure on main entry points, such as Italy and Greece. The number of arrivals in Italy showed little change for the entire period between 2014 and 2018, when it also dropped considerably. Each year from 2014 to 2017, an average of 150,000 people reached the Italian coasts smuggled by traffickers from North Africa and rescued at sea (UNHCR, 2018). In 2017, Italy received 67 per cent of the EU's migrant arrivals from Mediterranean routes and accounted for 18 per cent of all first-time applicants in the EU-28 (Eurostat, 2020).⁴

The Italian refugee crisis overlapped with the rise of right-wing populist parties with a clear anti-immigration agenda. The *Lega*'s populist radical-right message, in particular, has stoked nationalist and nativist sentiment, including attacks on immigrants, anti-globalization and Euro-sceptical views. After years on the fringe of Italian politics, *Lega* barged into power in the 2018 national elections, joining the major populist party, *Movimento 5 Stelle*.

In this paper, we assess whether the inflow of asylum seekers to Italy between electoral cycles led to a change in public support for radical-right anti-immigration parties and/or populist movements. We exploit the quasi-experimental setting provided by the Italian Dispersal Policy to examine

¹ The EU's border agency, Frontex, estimates that the number of unauthorized crossings on different routes across the Mediterranean, the Western Balkans, and Greece–Albania was about 10,000 per year from 2009 to 2013 before rising to 1.82 million in 2015. Between 2015 and 2017 Germany received the highest number of asylum seekers, followed by Italy and France (Eurostat, 2020).

²Throughout the paper, we use the term "asylum seeker" and "refugee" interchangeably. This is so as in our setting we cannot distinguish a person seeking asylum from one whose asylum status has been approved (i.e. a refugee).

³Aggregate refugee inflows dropped following the 2015 agreement between the EU and Turkey according to which migrants who do not apply for asylum or whose claims were rejected can be sent back to Turkey.

 $^{^{4}}$ In 2019 Italy received 9.3 per cent of the EU's migrant arrivals via Mediterranean routes and with 43770 application accounted for 6.3 per cent of all first-time applicants.

the causal effect of refugee migration on political outcomes in national Parliamentary elections. This policy was designed in 2014 by the Italian Home Office and involved quickly setting up temporary reception centers (CAS is the Italian acronym) to deal with the consistent arrival of asylum seekers and to cope with scarcity of the ordinary reception capacity run by local authorities. Unlike the latter, the CAS system is a private-enterprise system funded by the central government and managed by Italian Prefectures (government offices at the province level). According to the Dispersal Policy, the number of migrants allocated to each province is assigned based on the resident population, while the distribution of the centers within the provincial territory happens on a quasi-random basis through public procurement procedures.⁵ As a result of the significant and unanticipated dimension of the refugee crisis, CAS have hosted the vast majority of asylum seekers over the last few years in Italy such that from 'temporary facilities' they have been used as *de facto* first reception centers, contravening their exceptional nature. By leveraging natural exogenous variation in the share of refugees resettled across municipalities over the 5-year , we can overcome concerns about refugee sorting and estimate the short–run impact on electoral outcomes (vote shares) in national elections between 2013 and 2018.

For our analysis we use unique first-hand data on refugee resettlement and reception centers in the period from 2014 to 2019, collected by the authors through Freedom of Information Act (FOIA) requests to the universe of Italian Prefectures.⁶ We combine the above administrative dataset at the municipality level with data on electoral outcomes for national parliamentary elections from the Italian Home Office, and with municipality economic and demographic data from the National Institute of Statistics (ISTAT).

Our empirical analysis exploits the time and spatial variation in the share of asylum seekers allocated across municipalities induced by the design of the Dispersal Policy. This heterogeneity in the intensity of refugee reception is unanticipated and short-term in nature. The number of municipalities hosting a CAS tripled between 2014 and 2017, and the number of centers peaked between 2017 and 2018 (at the end of the observation period). The cross-sectional variation is

⁵The first dispersal programs were set up during the 1980s and early 1990s to manage refugee flows from Eastern Europe and the Balkans. These interventions were reinforced and upgraded all around Europe in response to the latest refugee crisis. The Italian Dispersal Policy (*Piano Nazionale di Riparto*) was designed at the height of the crisis in 2014, by creating 'temporary reception centers' or CAS (*Centri di Accoglienza Straordinaria*), in addition to the ordinary System for the Protection of Asylum Seekers and Refugees (SPRAR) run by municipal authorities. The CAS reception conditions are only 'basic' but they host between 75 to 80 per cent of asylum seekers arriving in Italy. We explain the Italian refugee reception system and the Dispersal Policy in details in Sections 2 and 3.

⁶This data on the refugee reception and relocation system is supposedly collected by the Italian Home Office from Italian Prefectures on a regular basis. Yet, this centralized database is not made publicly available. This is also the reason why a systematic impact evaluation of the reception system in Italy has been prevented thus far.

rather granular as it derives from the spatial dispersion of asylum seekers across almost 8000 small administrative units (LAU 2 - municipalities), with each CAS hosting between 20 to 35 refugees on average.⁷ Importantly, we provide evidence in support of the quasi-random design of the dispersal policy and show that both the level and the change in refugee allocation is not explained by systematic (political or economic) factors. As such, the Italian Dispersal Policy provides a clean setting to elicit the causal effect of hosting asylum seekers on voting behavior at the local level.

We find a positive effect of the share of asylum seekers on support for right-wing populist parties, although it is small in magnitude. A 1 percentage point (p.p.) increase in the share of asylum seekers increases the share of votes for anti-immigration parties by 0.24 p.p.. This corresponds to about 2.5 percent of total variation in vote shares. This effect is robust across dispersal policy characteristics and is not driven by the somewhat uneven distribution of large refugee centers (i.e. dispersion vs. concentration of asylum seekers). Yet, the political backlash appears to be heterogeneous across municipalities with different population size. The point estimate of the effect on anti-immigration voting among municipalities in the top 5 percent of the population distribution is lower than the benchmark estimate by about 3.8 p.p for both the Chamber of Deputies (*Camera* dei Deputati) and the Senate (Senato). However, the latter effect is not associated to an increase in left-wing electoral support, but is mainly explained by the electoral turnout, which is negatively affected by the share of asylum seekers in large municipalities compared to small ones. Moreover, we find that the anti-immigration backlash to asylum seeker inflows is attenuated in municipalities with a relatively high percentage of skilled people (i.e. with university degree or more) and exacerbated in municipalities with higher shares of (regular) immigrants. Finally, we show that results are not driven by greater economic hardship (related to the recent 2008 economic crisis) as the anti-immigration backlash is significantly lower in municipalities with a higher unemployment rate in 2011 (while the latter effect is associated with an increase in vote shares for the major populist party, M5S).

In order to assess in more detail the underlying mechanisms behind the increased support for antiimmigration parties, we next investigate the major economic consequences of refugee settlement at the municipality level. We find little economic impact, as we document that refugee settlement does not generate economic losses in terms of local average income, native outflows, and municipality public expenditure. In contrast, we observe that the dispersed allocation of asylum seekers, combined with the age structure of the local population, significantly contributes to the intergenerational replacement of elderly natives within municipalities.

⁷Our sample includes 6,965 municipalities with a mean (median) population of 7,729 (2,443) residents and a mean (median) area of 36 (21) square kilometres, each belonging to one of the 92 provinces covered by our sample.

Overall, we find little evidence that electoral preferences are rooted in actual economic mechanisms, leaving space for ideological or political drivers of voting behavior. Indeed, identity politics have played an increasing role in most European countries over the period under study, and it has become particularly divisive in Italy where fear of immigrants – and of anti-immigrant backlash – has overshadowed election discussions about the economy ahead of the vote on March 4th 2018. Hence, we investigate the role of political propaganda, occurred just before elections and right at the peak of the refugee crisis, in mobilizing voter support. We focus on the electoral campaign of right-wing parties, which has been significantly characterised by anti-immigration stances and openly xenophobic views.⁸ Using the Twitter accounts of right-wing candidates over the electoral campaign, we geocode information on right-wing political rallies that took place at the municipality level between January and March 2018. We show that while the right—wing campaign is fairly balanced across municipalities according to the share of hosted refugees, the impact of the latter on radical right-wing support is significantly higher (by about 1 p.p.) in areas where campaign rallies were held. Moreover, we combine this data on electoral rallies with the candidate Hate Barometer, collected by Amnesty International Italy during the three weeks before elections in order to measure political candidates' hate speech, i.e. speech against immigrants and ethnic minorities. We find that the anti-immigration backlash is significantly higher in municipalities visited by far-right candidates using hate speech. Overall, this is evidence of the role of the 'dog-whistle' effect, according to which radical-right propaganda may trigger electoral gains by legitimising some implicit stereotypes (e.g. fear of alien groups) through divisive language and stances (Goodin, 2008; López, 2015).

Our paper contributes to a growing body of literature studying the impact of refugee inflows on voting behavior and economic outcomes in host countries.⁹ In a recent influential work, Dustmann et al. (2019) examines the Danish refugee dispersal policy that assigned refugee immigrants across local municipalities on a quasi-random basis for more than a decade (1986–1998). By assessing the political impact over three distinct electoral cycles of both parliamentary and municipal elections, they find that electoral outcomes significantly vary according to pre-policy municipal characteristics and across rural–urban areas. In particular, while refugee allocation increases the vote share for anti-immigration parties in rural municipalities, the opposite holds in urban municipalities,

 $^{^{8}}Lega$'s political campaign slogan was 'Italians first' (*Prima gli Italiani*) and its leader, Matteo Salvini, pushed a political narrative focused on immigrant invasion, terrorism and threats to national security, identity, and the sovereignty of Italians.

⁹This literature builds upon a long–standing body of works studying the socio–economic consequences of immigration, which include among others Card (1990), Hunt (1992), Friedberg (2001), Borjas (2017), Peri and Yasenov (2019), and Clemens and Hunt (2019).

most likely because of different backgrounds or experiences with immigrants in the two subpopulations.¹⁰ By focusing on Upper Austria, where many municipalities hosted asylum seekers while also being exposed to transiting refugee flows to Germany, Steinmayr (2020) investigates the asymmetric impact of different forms of refugee exposure and 'inter-group contact' on voting for far-right parties in the 2015 state elections. The analysis employs an instrumental variables strategy to deal with the endogeneity in the distribution of asylum seekers across municipalities and finds that exposure to transiting refugees increased far-right votes, whereas refugee reception (and contact) with asylum seekers decreased radical-right support. Other recent works investigate the impact of refugees and immigration on the significant electoral gain of far-right parties across Europe in the last few years. For example, Edo et al. (2019), Lonsky (2021), Levi et al. (2020), Hangartner et al. (2019) and Gessler et al. (2019) examine anti-immigration backlash in France, Finland, Britain, Greece, Hungary respectively.¹¹ The direct effect of immigration on electoral outcomes has also been studied in the United States, where recent political leaders' positions on immigration have been shaping their own electoral success. By using and instrumental variable approach, Mayda et al. (2018) empirically assess the link between immigration and the vote share received by the Republican Party in the U.S. from 1990 to 2016. They find a positive and significant impact of low-skilled immigrants on the vote share for Republicans and a negative and significant impact of high-skilled immigrants. Their results are also consistent in showing that perceived costs and benefits of immigrants may differ according to local characteristics.¹²

By leveraging the high granularity of our refugee allocation data in Italy and the clean experimental setting whereby a sudden and massive inflow of asylum seekers is quickly dispersed across a country just before national elections, our findings complement the above evidence in isolating the causal impact of mere refugee exposure (i.e. without significant inter–group contact or any particular

¹⁰While using data from Denmark, Foged and Peri (2016) follow the effect on labor market outcomes of lowskilled natives between 1991 and 2008 using the exogenous inflow of low-skilled immigrants/refugees as identification strategy. They find that a larger supply of refugees pushed less educated native workers towards less manuallyintensive occupations, with positive implications for native unskilled wages, employment and occupational mobility.

¹¹See also Gehrsitz and Ungerer (2017), Albrecht et al. (2020), Freddi (2019) and Rozo and Vargas (2020) for further evidence on recent refugee exposure and political/behavioral outcomes in Germany, the Netherlands, Sweden and Colombia respectively. Before the recent refugee crises, other papers investigated the political impact of immigration while using an IV (shift-share) estimator to predict immigrant stocks based on historical settlement. For example, Otto and Steinhardt (2014), Mendez and Cutillas (2014), Barone et al. (2016), Halla et al. (2017) provide evidence on voting outcomes in Germany (Hamburg), Spain, Italy and Austria, respectively.

¹²They observe that while low-skilled immigrants have a pro-Republican effect in all counties, this effect is stronger in non-urban counties, counties with a high share of low-skilled natives and counties with a high share of public spending. High-skilled immigrants, instead, have a pro-Democrat effect in all counties but this effect is stronger in counties with a high ratio of public spending relative to income and greater linguistic diversity between natives and immigrants.

bundle of integration services) on the *rise* of right-wing populism and on the *change* in the whole political spectrum. This adds to the literature on the contentious drivers of different forms of populism and, more specifically, on the role of immigration shocks in fuelling the support for nationalist and nativist political movements (Rodrik, 2020; Guriev and Papaioannou, 2020; Alesina and Tabellini, 2020). Moreover, by exploiting our natural experiment whereby the peak of the refugee crisis occured during a political campaign, we can shed light on the mechanisms behind the anti-immigration backlash, and explore actual economic channels as well as non-economic mechanisms, such as the role of political propaganda. Indeed, it has been argued that identity politics and divisive political stances have been playing a growing role in the radicalization and polarization of voting behavior, with right-wing political campaigns shifting away from economic policy towards ethno-nationalist and cultural cleavage from which they derive their appeal (e.g. Gennaioli et al., 2019; Coates, 2017). While far-right parties have gained significant electoral success in Europe in recent years, there is evidence that they do not merely reflect, but also aggravate, radicalization, xenophobia and social conflict (see Bursztyn et al., 2019; Fletcher et al., 2020; Grosjean et al., 2020; Müller and Schwarz, 2018; Romarri, 2020). By measuring far-right political propaganda and hate speech during the electoral campaign in Italy, we show that antiimmigration propaganda significantly exacerbates the public opinion backlash.

The rest of the paper is organized as follows. Section 2 provides a description of the background and context. Section 3 presents the data and descriptive statistics. Section 4 describes our empirical strategy. Section 5 discusses our findings and presents the analysis of mechanisms and robustness checks. Section 6 offers some concluding remarks.

2 Background

2.1 The Refugee Crisis in Italy and the Reception System

Over the past twenty years, and even longer, Italy has turned into a major destination country for international migrants, either for permanent or transitional settlement, due to both domestic and external reasons, including the geographical accident of being a peninsula in the middle of the Mediterranean. After the arrival of a first large migration inflow during the 1990s, in particular from Albania and the Balkans following the Yugoslav Wars, there has been a steady South-North inflow to Italy from 2000 to 2007. Surges occurred afterwards due to the Enlargement of the European Union (2007), the First North African Emergency (2008), the Arab Springs, the end of the Libya regime and the subsequent exodus via the Mediterranean (2011), and the escalation of old and new conflicts in many areas of the Near East, especially Syria (2013-2014).

The so-called 'refugee crisis' in Europe peaked in 2015, with over 1 million asylum seeker arrivals in one year, mainly fleeing the Syrian war (UNHCR, 2018). The number of asylum seekers arriving in Italy, especially through the Central Mediterranean Route, showed little change between 2014 and early 2018. In 2015, 2016 and 2017, 155,000, 180,000, and 119,000 asylum seekers arrived in Italy respectively, smuggled by traffickers from North Africa or rescued at sea (UNHCR, 2018). This flow started declining in the wake of the Italy-Libya Memorandum of Understanding in 2017 and the right-wing populist government installed in 2018, with 20,120 arriving by mid-September 2018 and half as many migrants in 2019 (see Figure 1, where election cycles are also displayed).¹³





The need to host and settle asylum seekers fostered the scaling up of the existing reception system in Italy, which is designed along two major stages/tracks. A preliminary phase is related to identification and assistance, which is conducted at major disembarkation sites ('hotspots') and major governmental centers such as CARA (*Centri di Accoglienza per Richiedenti Asilo*).¹⁴

¹³The major entry points to the EU are Italy, Greece, Malta, Hungary, Croatia/Slovenia, and Bulgaria. The response to this crisis was to introduce border closures, first between Turkey and the EU (Greece) in 2015 and followed shortly thereafter by the borders between Serbia and Hungary and between Turkey and Bulgaria.

¹⁴ A preliminary phase of first aid and assistance applies to operations conducted in centers set up in the major spots of disembarkation. First Aid and Reception centers (CPSA) were created in 2006 for the purposes of first aid

Secondary reception is carried out by the System for the Protection of Asylum Seekers and Refugees (SPRAR), which was put in place in 2002 following the first refugee inflows. A main feature of this system is that it is set up and managed upon the decision of municipality administrations (LAU–2 administrative units).¹⁵ Thus, the SPRAR system is run by local authorities on a voluntary basis and is not for profit. It is funded by the national government through money channelled to local municipalities and provides reception services such as language courses, psychological care, training and labor market integration programs. The SPRAR system is often singled out for its small scale organization, aimed at refugee integration in the local context.¹⁶ Yet, since municipalities' political orientation and administrators capacity determine the presence and distribution of SPRAR reception centers across the country, at the height of the refugee crisis (2014–2015) in December 2014, only 433 out of around 8,000 municipalities were hosting a SPRAR project (and only 700 in 2017).¹⁷ This proved to be insufficient to manage the unprecedented inflow of asylum seekers that began in 2014.¹⁸

and identification before persons are transferred to other centers and are now formally operating as 'hotspots'. The latter are typically located on the EU's external borders, where the registration, identification, fingerprinting and debriefing of asylum seekers take place. These operations should be carried out within 48 hours of the arrival of the asylum seeker (72 hours in exceptional circumstances). There are 6 'hotspots' in Italy, 4 in Sicily and 1 in the Apulia region. Afterwards, a first reception stage is implemented in existing collective centers or in centers established by specific ministerial decrees (e.g. CARA, CPR- see below) or, in case of insufficient space, in 'temporary' structures. Asylum seekers should be hosted here for a maximum of 35 days, while waiting for their asylum application to be processed. In reality though, many people end up staying much longer. At the height of the migration crisis in 2015 and 2016, CARA di Mineo (Catania Province in Sicily) was Italy's biggest reception center with more than 4,000 migrants. These big centers are usually characterised by huge facilities, high costs, low quality of services provided, and isolation from residential areas. CPR (Return Detention centers) are where immigrants may be detained if they refuse to be identified upon arrival, do not have a valid permit of stay or received an administrative expulsion order. The maximum duration of the administrative detention was 90 days in 2013. Recently, Art 2 of the Security Decree extended the maximum period of detention to 180 days. In Italy 6 CPR are currently active: Rome (Lazio). Bari and Brindisi (Apulia), Turin (Pindemont), Palazzo San Gervasio and Potenza (Basilicata), Caltanissetta and Trapani (Sicily).

¹⁵The administrative division of Italy includes 20 regions and each region is divided into a number of Provinces (corresponding to NUTS-3 administrative units), which in turn are divided into Municipalities (LAU-2). Overall, there are currently 107 Provinces. The average population size of provinces is around 540,000, ranging from 127,844 to 3,075,083. Provinces are made of municipalities, which number slightly less than 8,000 and have an average population of around 7,000 (ranging from 120 to more than 1 million).

¹⁶The SPRAR model provides for a joint action of the Ministry of the Interior, the National Association of the Italian Municipalities (ANCI) and the United Nations High Commissioner for Refugees (UNHCR) and is supported by shared responsibility between local and central public authorities. For detailed information see https://www.siproimi.it/english

¹⁷By using data on SPRAR centers between 2005 and 2017, Gamalerio (2019) shows that opening a center allows municipalities to receive substantial fiscal grants from the central government and to gain investment's returns for the local economy. Amid the economic benefits, he finds that electoral incentives deter local politicians from opening a SPRAR center.

¹⁸Moreover, the redistribution of migrants has occurred not uniformly everywhere, with municipalities opening reception centers at different points in time. By using an IV strategy based on the availability of group accommodation buildings as a predictor of SPRAR settlements, Gamalerio et al. (2020) evaluate the impact of the latter on

Hence, a third (parallel) track of (second stage) reception centers was set up on a 'extraordinary' basis. These Temporary Reception centers (*Centri di Accoglienza Straordinaria* - CAS) were created to deal with the lack of capacity of the existing reception centers and quickly replaced both SPRAR and CARA as the go-to system for the new government. Each year in the period of 2014–2018 CAS centers hosted around 75 per cent of asylum seekers in Italy (Def. 2018– see Figure A1 in Appendix). CAS is a private-enterprise system financially supported by the national government, but run by private stakeholders who provide services (food and accommodation) for asylum seekers and refugees. Some CAS are housed in former group accommodation buildings but the vast majority are divided across networks of private apartments.¹⁹

The number of asylum seekers is centrally allocated to province-based Italian Prefectures according to the "Allotment Plan" (*Piano Nazionale di Riparto*), which sets the number of asylum seekers as a proportion of the provincial resident population (i.e. about 2.5 out of 1,000 inhabitants).²⁰ The allocation of centers within the provincial territory is coordinated by local Prefectures, which open public bids that are eventually assigned to cooperatives, NGOs or private operators based on the quality of the project and the tender cost schemes.²¹ The location of refugee centers are proposed and decided by economic operators, without consultation with local municipality administrations. Within the terms of the law, procurement calls remain open for 35 days, even less if a case of urgency is made. At the height of the crisis, the vast majority of procurement calls were set up with the 'competitive open procedure' (i.e. any interested operator may submit an offer in response to a summons for bid) and unfilled bidding is less than 2 per cent (see ActionAid and Openpolis, 2018)²²

extreme-right voting. While using data from 2016, i.e. before the peak in refugee inflows, on both SPRAR and CAS together, Bratti et al. (2020) instead evaluate the geographical spillover effects of refugee premises on voting outcomes in neighbouring municipalities.

¹⁹As reported by Parlamentary Commission to Chamber of Deputies in 2017, 85 per cent of CAS centers are hosted in apartments or private houses, most of them (82 per cent) privately rented (Chamber of Deputies, 2017).

²⁰The *Piano Nazionale di Riparto* was drawn up in July 2014 for both SPRAR and CAS systems, and it was revised in December 2016, when the Home Office tried to facilitate the opening of SPRAR centers for municipal authorities. For instance, the "safeguard clause" was included, which exempts municipalities involved in the SPRAR network from other forms of reception, as long as they meet the above ratio (if some CAS were already in place, they would be downsized or converted into SPRAR centers). Other financial and administrative measures were introduced in order to incentivize local authorities to agree to join the ordinary system. Yet, this did not work in practice, as SPRAR centers grew by only 20 per cent in 2016 (primarily in Provincial capital), and this was not enough to keep pace with new arrivals. As a consequence, CAS grew threefold by 2017. The functioning of the reception system has been regulated by Legislative Decrees (LD) 142/2015 and 132/2018, which distinguishes among different stages of asylum seeker reception.

²¹A dysfunctional clash between Prefectures and local mayors regarding the allocation of asylum seekers and the opening of receptions centers often made headlines for leading to anti–immigration protests.

 $^{^{22}}$ Between 2016 and 2017 the amount tendered with the 'open procedure' grew by about 1 billion euros. This is the procedure in which the highest level of transparency is guaranteed and competitiveness between the various

This Dispersal Policy aimed to reduce the concentration of asylum seekers and refugees in urban and disembarkation areas, as well as share the "costs" of reception and hospitality through a gradual and sustainable distribution of asylum seekers across the whole national territory. However, a critical point of the CAS system is that, unlike SPRAR, its 'temporary' and the private nature does not aspire to ensure the provision of integrated refugee reception services such as psychological support, training and job market integration ²³. Thus, while CAS ended up being geographically dispersed, they provide fewer integration services than the ordinary system, such that very few asylum seekers in CAS can participate in the labor market, for example.²⁴ Finally, it is worth mentioning that Italy has the longest asylum procedure in Europe, ranging between 18 and 24 months from application to formal status registration.

Overall, the extent to which the Dispersal Policy has been efficient in finding a balance between dispersion and concentration, hence diluting costs and stemming 'not-in-my-back-yard' backlash, is not clear a priori. More evidence and descriptive statistics on policy design and CAS distribution across Italy is provided in Section 3.

2.2 The Political Spectrum in Italy

The outcome of the 2018 round of national elections brought about a radical transformation in the balance of power across the political spectrum. First, anti-immigration parties gained a sizeable growth in their share of votes and took the lead within the center-right area. Second, the populist *Movimento 5 Stelle* (M5S) became the most-voted for party in both chambers of national parliament. Third, center-left parties experienced a large loss of consensus among their former voters.

Within the anti-immigration front, two parties gained the most significant momentum: Lega and

economic operators involved is maximized (ActionAid and Openpolis, 2018).

 $^{^{23}}$ The Italian government used to assign 35 euros per person per day (mostly in goods and services) for both SPRAR and CAS. Yet, in the case of CAS, this an indicative figure as the actual costs are established through calls for tenders, and may therefore be lower. Moreover, as far as the CAS are concerned, each Prefecture in Italy can change the starting auction base. Critics of the predominance of CAS systems worry that since these centers are being run as an enterprise rather than solely to benefit the local community, the managers might be tempted to cut costs in order to turn a profit.

 $^{^{24}}$ Asylum seekers can potentially get a job two months after they apply for asylum. However, if their income reaches a certain threshold they become ineligible to be hosted in the reception centers. According to official statistics, less than 10 per cent of all asylum seekers hosted in the Italian reception system had a regular job contract between 2011 and 2017. See http://documenti.Chamber of Deputies.it/leg17/resoconti/commissioni/bollettini/pdf/2017/09/06

Fratelli d'Italia (FDI). The former started in northern Italy in the late 1980s as a federalist party promoting regional fiscal independence and more recently became a national party with the leadership of Matteo Salvini. The latter is a nationalistic party created by scission from the center-right party – Popolo delle Libertà (PDL) – and can be roughly identified with the post-fascist political area. The remainder of the anti-immigration front includes several extreme-right groups that are best exemplified by Casa Pound (CP), a neo-fascist movement embracing nationalistic ideas, such as nationality-based welfare systems, and repudiating the reception of asylum seekers in Italy.

Anti-immigration and nativist sentiments have been a salient element of both *Lega* and FDI's political agenda. Their propaganda has heavily emphasised the risk that migration may trigger a process of demographic and cultural change ('ethnic substitution'), and has depicted irregular migrants as a potential threat to the national economy and security. While proposing fairly similar economic programs, these parties share common stances on several migration-related issues, combined with anti–establishment and Euro-sceptical stances. In fact, both i) have not endorsed any reform of actual laws regarding regular immigration to Italy; ii) strongly opposed the reception of irregular migrants crossing the Mediterranean by sea; and iii) voted against the reform of the Dublin system in the European Parliament.²⁵ Importantly, the rise of *Lega* has given new life to the extreme right fascist ideology and fed a climate of racially motivated attacks has increased sharply over the last few years in Italy (Romarri, 2020).

In contrast, center–left parties have supported the change in the Dublin Regulation.²⁶ The *Partito Democratico* (PD) was the major component of the government that administrated Italy during the refugee crisis, when the CAS asylum-seeker reception scheme was introduced and implemented. Even though the PD-led government set up an agreement with Libyan authorities in 2017 that was effective in preventing a substantial portion of irregular flows from overseas, the PD was perceived as a pro-immigration party among electors. *Lega* and FDI representatives blamed the PD-led

 $^{^{25}}Lega$'s leader Matteo Salvini was appointed Interior Minister in the 2018 populist coalition government. Just 10 days after his appointment he started the 'no docking rights policy' for NGOs-run rescue boats in Italian ports. In September 2018 he implemented the controversial Security Decree, which denied asylum seekers the right to enrol in municipal registry offices anymore (*de facto* excluding them from basic health care and social coverage), weakened the rights of migrants and asylum seekers, excluded them as well as beneficiaries of humanitarian protection status from the SPRAR system, and downsized the CAS system by reducing financial support for general services, including mediation and legal advice to refugees.

 $^{^{26}}$ By center–left parties we refer to *Partito Democratico* (PD) and 'radical-left' parties, i.e. *Sinistra, Ecologia e Libertà* (SEL) in 2013, and *Liberi e Uguali* (LEU), in 2018. While SEL was part of the same coalition as PD in the 2013 election, LEU presented an independent coalition with its own candidate for prime minister in 2018. However, we include both them in the center-left coalition as they have very similar positions with regard to immigration policies.

government for allowing a massive inflow of irregular migrants.

Finally, M5S has rejected a precise placement over the left-right axis and has held an ambiguous position on immigration. Indeed, while not explicitly engaging in anti-immigration propaganda, M5S's European Parliament members voted against the reform of the Dublin Regulation on the distribution of asylum-seekers across European countries.

In order to document the ideological differences on immigration across Italian parties, in Table 1 we report data from the Manifesto Project (Volkens et al., 2020). The latter extrapolates, through text-analysis of political manifestos, election-specific information about parties' positions on a large range of issues. None of the manifestos by major parties, except PD in the 2018 elections, mention cultural diversity as a desirable feature of society. In fact, the manifestos from both *Lega* and FDI include negative references to diversity. Aversion to multiculturalism became prominent for these parties after the breakout of the refugee crisis. For instance, while absent in the program for the 2013 election, the appeal for cultural homogeneity and against the risks of a diverse society became a stable pillar of *Lega*'s program in 2018. The political programs of all right–wing parties for the 2018 elections, and in particular *Lega*, include statements in favour of restrictions to immigration, while only PD's manifesto conveys a positive view on this subject.²⁷ The manifestos by *Lega* and FDI, moreover, demand a process of integration for immigrants who should be hence expected to fully assimilate into national culture, rather than retain their own customs and cultural traits.

			2013					2018		
Category:	Lega	FDI	PDL	M5S	PD	Lega	FDI	PDL	M5S	PD
Multiculturalism: Positive	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.23
Multiculturalism: Negative	0.00	1.52	0.00	0.00	0.00	2.75	3.85	0.00	0.00	0.00
Immigration: Negative	-	-	-	-	-	2.98	1.65	2.08	0.02	0.00
Immigration: Positive	-	-	-	-	-	0.00	0.00	0.00	0.09	0.39
Immigrants Assimilation	-	-	-	-	-	1.91	2.20	0.00	0.00	0.00

Table 1: Manifesto Project dataset - Italian parties' ideology about immigration

 $^{^{27}}$ The Manifesto Project's data on party ideology concerning immigration and assimilation are not available for 2013. We therefore cannot document the within-party ideological evolution of these subjects between 2013 and 2018. Data source: https://manifesto-project.wzb.eu/

3 Data and Descriptives

3.1 The Dispersal Policy

Open access to centralized data on CAS and refugee distribution across Italy from 2014 onwards is not made available by the Home Office. Hence, we carried out first-hand data collection of administrative information through Freedom of Information Act requests (FOIA - Accesso civico generalizzato) to governmental offices (Prefectures) at the local (provincial) level.²⁸ We filed formal requests for data access to the universe of 106 Prefectures between July 2019 and February 2020.²⁹ We collected information on CAS capacity, timeline and actual number of hosted refugees for the years 2014–2019.³⁰

The format of the data received and the response times were very heterogeneous across offices. In most cases, the data contained the list of reception centers set up in a province area within the reference period along with details on location and capacity. We obtained complete data for the entire period (2014-2018) for 92 Prefectures. In 10 extra cases, data were made available only for the most recent years, so we could not include those provinces in our analysis. Four offices did not release any data at all.³¹

Information was extracted through text mining and machine learning techniques from very different source files and combined into a unique and harmonized dataset. The compiled dataset allows mapping the number of asylum seekers hosted across 6,965 out of 7950 Italian municipalities (7,918 in 2019) and tracking their evolution over time from the escalation of the refugee crisis to date (2014-2019). Overall, our sample provinces hosted around 37,000 asylum seekers in the CAS reception system in 2014, which increased to almost 144,000 in 2017 (equal to about 0.3 per cent of the total national population). In Figure 2 we show a map of our data for 2014 and 2017, namely the share of asylum seekers on the 2013 resident population across Italian municipalities.

 $^{^{28}}$ The Italian Home Office has a local representative unit (i.e. Prefectures) in 106 of them. Sardinia has 4 local offices covering a territory of 5 Provinces.

²⁹In order to test the response rate and time we firstly ran a pilot survey with 23 local offices from three regions (July-September 2019). As a second stage, data collection was extended to the remaining 83 offices. A description (in Italian) of our data collection is available at https://www.openpolis.it/limportanza-dellaccesso-ai-dati-il-caso-del-sistema-di-accoglienza-in-italia/

 $^{^{30}}$ For a subsample of Provinces we have additional information on the characteristics of hosted refugees (gender, number of unaccompanied children, nationality) and the type of accommodation and management facility.

³¹Palermo, Campobasso, Isernia and Cremona never replied FOIA requests and are excluded from our sample. As for response time, this varied between two weeks and five months. In more than half of cases, several interactions between the offices and the research team were needed to obtain the requested data.



Figure 2: Share of asylum seekers in the municipality population 2014-2017

The Refugee Dispersal Policy implemented by the Italian government reproduces a quasiexperimental setting, by assigning refugees to municipalities on a quasi-random basis. As shown in Figure 3, where we plot province population size before the policy launch (2013) against the number of asylum seekers, the latter are first allocated to each province based on the resident population. Provinces with larger populations received larger numbers of refugees on a mandatory basis. The regression line in Figure 3 has a slope equal to 0.0023 (s.e. 0.0001), which is very close to the Allotment Plan of 2.5 asylum seekers out of every 1,000 inhabitants. The strong relationship between allocated asylum seekers and pre–policy population size is also reflected in the regression goodness-of-fit measure (R–squared equal to 85 per cent). The latter relationship is naturally reciprocated at the municipality level, although with much more variation in the allocation numbers across similarly–sized municipalities. Figure 4 plots the allocated number of refugees against the (pre–policy) municipality population size and, while the regression slope is similar to the former one (0.0024 with s.e. 0.00003), population size explains only 47 per cent of the variation in the number refugees. This variation is the result of the within-province refugee allocation process coordinated by Prefectures through the public procurement bids we described in Section 2. In

Figure 5 we plot the share of asylum seekers against some municipality characteristics at baseline such as the average (log) income per capita, unemployment rate, population share with university degree, and share of foreign population.³² Plots show no systematic correlation between the refugee share and municipality–level characteristics. We provide more evidence on the quasi-random allocation of refugees across municipalities below, when describing the identification strategy below. Moreover, beyond cross-sectional variation, in our analysis identification is achieved by exploiting within–municipality variation in the allocation of refugees between the two electoral cycles.

We report descriptive statistics in Table 2. The number of municipalities hosting a CAS tripled over time along with the increase in asylum applications. The maximum number of CAS centers was reached between 2017 and 2018 (at the end of our observation period). Reception centers host between 20 and 30 refugees, on average, with a decreasing average size over time as long as the dispersal policy has been put into practice. Yet, high heterogeneity is observed in the organization of hospitality at the local level. For instance, given the same percentage of asylum seekers in the total population, the average number of refugees per CAS centers in 2017 was 16 and 101 in Lombardy and Sicily, respectively. Overall, the percentage of reception centers hosting more than 100 refugees never exceeded 3.5 per cent. This evidence reflects the granular level of the dispersal policy implemented in Italy, which was the result of a scramble for a rapid geographical allocation of asylum seekers.



Figure 3: Number of asylum seekers hosted in 2017 on 2013 province population

³²Baseline data are for 2013 or 2011 depending on availability.



Figure 4: Number of asylum seekers hosted in 2017 on 2013 municipality population

Figure 5: Asylum seekers allocation and municipality characteristics



	2014	2015	2016	2017	2018	2019
Municipalities hosting a CAS	775	1383	2135	2655	2340	1893
Municipalities hosting a CAS $(\%)$	11.13	20.43	31.50	38.12	35.01	27.18
Nr. of asylum seekers	37374	67566	121258	143750	116014	83690
Nr. of CAS	1786	4539	10187	15223	15948	10743
Average size of CAS	34	28	26	23	22	20
CAS with more than 100 guests	62	119	213	254	191	152
CAS with more than 100 guests $(\%)$	3.5	2.6	2.1	1.7	1.2	1.3

 Table 2: Allocation of asylum seekers

3.2 Electoral Data

In order to measure local political preferences, we use data from the Italian Home Office for the 2008, 2013, and 2018 national elections of the Chamber of Deputies and Senate³³, the two chambers of the Italian Parliament.³⁴ This dataset provides information about the number of votes received by any party competing in each round at the municipality level.

At a national level PDL, the major holder of center-right governments in the last 25 years, accounted for about 22 percent of total votes in the 2013 national elections, while *Lega* and FDI, combined, accounted for around 6 percent. After the March 2018 elections, the anti-immigration front reached the leadership within the center-right coalition. FDI more than doubled its share of votes, and *Lega* received around 17 percent of votes (see Table 3). In the Appendix, we report a map of vote shares for anti-immigration parties at the municipality level (see Figure A2). The results from the 2018 elections also confirmed the rising trend in the support for the populist party M5S, which escalated to about 32 percent of preferences from, 25 percent in 2013.

Between 2013 and 2018, leftist parties experienced a drop in their consensus. Center-left coalitions received, on aggregate, about 26 percent of votes in 2018, down from around 30 percent in 2013. The *Partito Democratico* (PD) maintained the lion's share, with slightly more than 18 percent and 19 percent of votes for, respectively, the *Camera* and *Senato* elections, although it reported the relatively largest loss if compared to 2013, when PD accounted for more than 25 percent. The second largest parties within the left wing of parliament were *Sinistra, Ecologia e Libertà* (SEL), in 2013, and *Liberi e Uguali* (LEU), in 2018. Both are located to the left of PD on the political spectrum, and their share of votes amounted to around 3 percent in both elections.

 $^{^{33}\}mathrm{We}$ consider national elections in 2008 to test for diverging pre-trends in political preferences during pre-refugee–crisis years.

³⁴Data are publicly available at: https://elezionistorico.interno.gov.it/.

We examine the impact of asylum–seeker presence on the share of votes for the whole group of antiimmigration parties, including *Lega*, FDI and CP. We further assess the same effect separately for the two main anti-immigration parties, i.e. *Lega* and FDI. We further estimate vote shares for the center-right party (PDL) and the populist party (M5S). We then focus on the group of center-left parties, including the 'Democratic Party' (PD) and 'radical-left' parties (i.e. SEL in 2013 and LEU in 2018). Finally, we consider political participation by using the municipality's electoral turnout (i.e. the share of actual voters over the number of citizens entitled to vote) as an outcome.³⁵

	2013		2018	
	Chamber of Deputies	Senate	Chamber of Deputies	Senate
Anti-immigration parties	6.05	6.25	21.70	21.87
Lega	4.09	4.33	17.35	17.61
FDI	1.96	1.92	4.35	4.26
PDL (Forza Italia)	21.56	22.30	14.00	14.43
M5S	25.56	23.79	32.68	32.22
center-left	29.55	31.63	26.25	26.28
Turnout	75.19	72.93	75.11	72.99

 Table 3: Election outcomes - Descriptives

Notes: Source: Home Office data warehouse - Electoral results.

3.3 Political Propaganda

To measure the political propaganda of anti-immigration parties before the 2018 elections at the local-level, we use an indicator of public events and rallies of right-wing parties at the municipality level. We gather this information from Twitter accounts belonging to Lega, FDI and PDL candidates, as well as official party accounts, and geo-reference local events or electoral rallies happening across Italian Municipalities between January and March 2018. We do so in order to leverage the high granularity of exposure to political propaganda and to measure right-wing stances delivered in person and not on-demand as is the case with media access or engagement. In fact, electoral campaigns in Italy are rally-intensive, and particularly so in the 2018 election, when identity politics played an important role and 'charismatic politicians', such as the *Lega* leader Salvini, exploited the traditional ground campaigning and voter contacts to influence others.

We downloaded more than 42,600 tweets published by election candidates and official right-wing

 $^{^{35}}$ All Italian citizens over 18 years old are entitled to vote for the election of the members of the Chamber of Deputies, while only those over 25 years of age are entitled to vote for the election of Senate members.

party accounts from January, 1st to March, 4th, the day of national elections. By using an automated algorithm, we gather all tweets that reference any election event (rallies, electoral meetings, political public event, etc.) held in a specific location (municipality).³⁶ We found 4,300 tweets linked to an electoral campaign event at the local level, and among them we identify 851 municipalities that has been visited at least once by right–wing candidates during the final rush of the electoral campaign (see Figure A3 in Appendix). About half of these municipalities happened to also host a CAS in 2017.

To further elicit the 'dog-whistle effect' of anti-immigration propaganda on orienting electoral preferences, we use information collected by Amnesty International Italy in their 'Hate Barometer' (*Il barometro dell'odio*).³⁷ Amnesty monitored social network (Facebook and Twitter) profiles of all candidates to both Chamber of Deputies and Senate during the 23 days before the 2018 election, reporting the online hate speech episodes of incitement to hatred against immigrants and ethnic minorities. Overall, they collected 787 episodes (either Facebook posts or tweets) from 192 candidates, 82 per cent of them from the main anti-immigration parties. The vast majority of hate speeches (91 per cent) were directed against immigrants and foreigners. From these records, we identify 125 right-wing candidates who published at least one hate message during the observed period. We match this information with the dataset on electoral campaign events described above and we single out 36 hate-speech-candidates who also held a rally in 231 municipalities across the country during the electoral campaign.

3.4 Municipality characteristics

We match the source of data described so far with a set of demographic and economic characteristics of the municipalities. We use these variables to check whether the allocated share of asylum seekers is independent of a set of observable local-level variables in the baseline period, and to investigate the mechanisms driving our empirical findings. We consider data from ISTAT's public warehouse³⁸ on the resident population by age and gender, the share of foreign born population, the share of the population over 65 years old and per–user expenditure for local public services. We also use

³⁶The algorithm includes the use of keywords such as 'rally' 'meeting' 'gathering' etc. Such keywords must appear in combination with the name of an Italian Municipality, which we then match to our main dataset. A final, semi–automated procedure is used to double–check if all tweet–municipality pairs actually identify a political meeting.

 $^{^{37}{\}rm For}$ detailed information on Amnesty International report see https://www.amnesty.it/barometro-odio/ (in Italian

³⁸Available at: http://dati.istat.it/.

data from the latest available 2011 Census on municipality unemployment rates, share of the population with tertiary-education, and presence of non-profit institutions.³⁹ Finally, we resort to aggregate data from the Minister of Finance on taxable gross income earned by residents to compute municipality per-capita income.⁴⁰ The 6,965 municipalities in our sample have a mean (median) population of 7,729 (2,443) residents, ranging from 120 to more than one million, and a mean (median) area of 36 (21) square kilometres. The mean (median) per capita income is 11,686 (12,165) Euros and the mean (median) share of residents over 65 years is 23 (22.40) per cent. The mean (median) share of foreign residents before the refugee crisis is 6.21 (5.37) per cent.

4 Empirical Strategy

We study the impact of the presence of asylum seekers on electoral outcomes at the municipality level by means of a fixed effects model specified as follows:

$$V_{mt}^{j} = \alpha^{j} + \beta^{j} A S_{mt} + \mu_{m}^{j} + \delta_{t}^{j} + \epsilon_{mt}^{j}$$

$$\tag{1}$$

The dependent variable, V_{mt}^{j} , is the vote share (over the total number of voters) for political party (or group of parties) j in municipality m at time t (where $t_0 = 2013$ and $t_1 = 2018$). Our analysis considers the outcomes from the 2013 and 2018 national elections for the two chambers of the Italian Parliament, i.e. the Chamber of Deputies and Senate. The explanatory variable of interest, AS_{mt} , is the share of asylum seekers at the municipality–level (i.e. the number of allocated refugees as a fraction of the municipality's total population at the baseline).⁴¹ This is computed as the sum of the capacity of all CAS in a municipality. As we consider the allocated number rather than the actual number of refugees living in a municipality, the coefficient β can be interpreted as an intention-to-treat parameter (see also Dustmann et al., 2019). Since the CAS system started in 2014 (the crisis kick off), and before then refugee reception in extraordinary centers was either low or nil at the local level, the share of asylum seekers is set to 0 in the pre–treatment period (2013)

³⁹Source: http://dati-censimentopopolazione.istat.it/Index.aspx?lang=it.

⁴⁰Source: https://www1.finanze.gov.it/finanze3/stat_dbNewSerie/index.php.

⁴¹While we have no precise data on refugee administrative registrations, we know that some municipalities registered refugees hosted in CAS centers among the resident population, while others did not (registration is generally the rule though, since it allows asylum seekers to get access to basic health and social services). This registration issue may generate inconsistencies in the population size across municipalities depending on the allocated number of asylum seekers. For this reason, we standardize the number of asylum seekers with total population size in 2013, i.e. just before the CAS system was introduced.

in all municipalities, while in $t_1 = 2018$ it is equal to the refugee share in 2017, the year just before the elections.⁴² ⁴³

The parameter μ_m captures municipality fixed effects and all time-invariant characteristics at the local level, while δ_t , the parameter for time fixed effects, accounts for shocks that are common to all observations in a given year. Importantly, municipality fixed effects absorb any static determinant of voting behavior including the local historical presence of *Lega (Nord)* or extreme right parties at the municipality–level, as well as cross-sectional variation in the duration of refugee reception, the geographical municipality area, local infrastructure, cultural traits, and social norms. ϵ_{mt} is an idiosyncratic error component. Standard errors are clustered at the municipality level.

The identification of β as the causal effect of the share of asylum seekers on local political preferences hinges on the assumption that treatment allocation - the share of asylum seekers – is independent from any unobserved local feature, embodied in the error term ϵ_{mt} , that affects the outcome at the same time. If asylum seekers' allocation is correlated with municipality characteristics at the baseline that simultaneously affect local political preferences, and if the error term ϵ_{mt} is serially autocorrelated, the estimate of the causal impact will be spurious. To illustrate, assume for example that municipalities with higher exposure to asylum seekers perform differently in economic terms before the program launch. Then, preferences for anti-immigration parties may increase through channels not related to exposure to asylum seekers, and the estimation of causal effect will be biased upward.

In order to check whether local characteristics at the baseline are associated with the intensity of the treatment, we first run a balance test of the change in the share of allocated asylum seekers between 2013 and 2017 on a set of municipality-level variables in the pre-treatment period. Table 4 reports the results for balance tests. Each entry represents the outcome from the univariate cross-sectional regression of the share of asylum seekers on each local pre-treatment variable. Results reveal that the treatment is fairly balanced across almost all of the observables we consider, i.e. economic, demographic, political and local environment characteristics. The endline share of asylum seekers is not significantly correlated with (log) per capita income, share of foreign borns and (log) per user welfare expenditure in 2013, and with unemployment rate and share of population with college degree in 2011 (the last available year). There is also no correlation with the opening of a SPRAR center over the same period, municipality being under receivership in the former period

 $^{^{42}}$ The 2018 Italian general election was held on March 4th 2018 after the Italian Parliament was dissolved by President Mattarella on December 28th 2017.

 $^{^{43}}$ In a set of robustness checks, we use the maximum share of asylum seekers hosted at any point in time between 2014 and 2017 as our endline observation. Results are qualitatively the same (available upon request).

(as a proxy for the local level of corruption), number of NGOs and local volunteers (civic capital). Interestingly though, the share of asylum seekers is significantly and positively associated with the share of the population over 65 years old, i.e. municipalities with a higher fraction of elderly residents seem to receive, on average, more asylum seekers than municipalities with a lower share of elderly population. We will consider this throughout the analysis.⁴⁴

As for political variables, the share of asylum seekers is not significantly associated with the share of votes for anti-immigration parties and PDL in the 2013 election for the *Chamber of Deputies*. Yet, it is (weakly) positively associated with the share of votes for PD at the municipalities level. As we will discuss along with our results, this weak imbalance cannot explain our findings but it can eventually lead to a lower-bound interpretation of our estimates.

We also examine whether the allocation of asylum seekers in 2017 is independent from pretreatment trends in local political preferences and income per capita. More specifically, we estimate the same model as in Equation (1) but considering the variation in the outcome between 2008 and 2013. In the same spirit as a placebo test, we assign to observations in 2013 the value of the share of allocated asylum seekers in 2017 and 0 for all municipalities in 2008. It is worth noting that the political spectrum changed between 2008 and 2013, and new parties that did not exist in 2008, e.g. FDI and M5S, emerged and gained consensus. Thus, we can test for parallel pre-trends for parties present in both 2008 and 2013 only. The results in Table 5 show that the share of allocated asylum seekers in 2017 is not significantly associated with within-municipality changes, between the 2008 and 2013 elections, in the share of votes for Lega (Column 1), the most voted anti-immigration party. However, we find that the share of asylum seekers afterwards is very similarly correlated with the growth of the vote shares between 2008 and 2013 for PDL and PD, the main parties of the center-right and center-left coalition respectively (Columns 2 and 3). These results point to the fact that asylum seekers were distributed 'equally' with respect to previous political trends, in accordance with the dispersal policy design. Yet, diverging pre-trends in political outcomes may affect estimates of the causal effect in Equation (1). Indeed, if we find, that the share of asylum seekers between 2013 and 2018 is positively associated with the share of votes for PD or PDL, we will not be able to fully distinguish between the causal impact of the

⁴⁴This is the result of the dispersal policy whereby asylum seekers do not choose where to live, while (young) natives do. Differently said, while asylum seekers are dispersed across municipalities, the age-pyramid is not. This is related to the demographic structure and geographical distribution of the Italian population. While the latter is ageing (22.8 percent of the total population was 65 years old in 2019, the oldest population in the EU) and elderly people are relatively spread out, young people are more geographically concentrated (e.g. in metropolitan areas). Hence, the latter concentration contrasts with the dispersion of asylum seekers, which mechanically mimics the dispersion of elderly people.

treatment from ongoing trends in the dependent variable. However, we show this is not our case in the discussion of results.

Finally, the estimate in Column 5 does not yield a significant correlation between asylum seeker allocation and pre-treatment trends in income per capita. Municipalities hosting a higher share of asylum seekers were not therefore on a different economic growth trajectory in the years before the refugee crisis.

Baseline municip. characteristics	AS share (endline)	N
Economic and demographic characteristics		
Log(Income per capita)	-0.0414	6965
	(0.0855)	
Unemployment rate	-0.0028	6896
	(0.0037)	
Log(welfare exp. per user)	-0.0466	6965
	(0.0358)	
%. univ. degree	-0.0083	6965
-	(0.0080)	
% over 65	0.0168***	6965
	(0.0043)	
% of foreign born	-0.0047	6965
	(0.0044)	
Political characteristics		
Anti-immigration parties	-0.0001	6965
<u> </u>	(0.0027)	
Lega Nord	-0.0004	6965
0	(0.0028)	
FDI	0.0017	6965
	(0.0063)	
PDL	0.0011	6965
	(0.0030)	
M5S	-0.0057	6965
	(0.0039)	
center-left	0.0049^{*}	6965
	(0.0025)	
Election turnout	-0.0036	6965
	(0.0030)	
Local institutions and civic/social capital		
Municipality hosted a SPRAR	-0.0544	6965
	(0.0534)	
Municipality under receivership 2007-13	-0.0474	6965
- •	(0.0409)	
% of NGOs serv. soc. emerg. p.c.	0.3562	6965
	(0.3192)	
% of NGOs int. coop. p.c.	0.1180	6965
	(1.5423)	
% of NGOs religion p.c.	0.7863	6965
Ŭ.	(0.8617)	
% volunteers serv. soc. emerg. NGOs p.c	0.0094	6965
	(0.0140)	
% volunteers int. and coop. NGOs p.c	-0.0122	6965
	(0.0223)	
% volunteers religious NGOs p.c	0.0051	6965
~ •	(0.0034)	

Table 4: Balance tests on pre-treatment municipality characteristics

Notes: Standard errors clustered at the province level in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the change in the share of asylum seekers hosted in the municipality between 2013 and 2017 (where the baseline is equal to zero) over the 2013 total population. Column 2 reports the coefficients of the regression of AS share on pre-treatment variables. (Log) income per capita, share of over 65 and foreigners, SPRAR (dummy), per-user welfare expenditure and electoral outcomes refer to 2013; unemployment rate, % of population with tertiary education and % of NGOs and volunteers on total population refer to 2011.

				1 1	
	(1)	(2)	(3)	(4)	(5)
	Lega Nord	PDL	PD	Turnout	Log(Income p.c.)
	(Chamber oj	f Deputies		
AS share	-0.0169	0.0977^{**}	0.1027^{**}	-0.0931**	0.0004
	(0.0483)	(0.0460)	(0.0459)	(0.0369)	(0.0004)
		Sen	ate		
AS share	-0.0547	0.1111^{**}	0.1128^{**}	-0.0981^{**}	
	(0.0448)	(0.0456)	(0.0451)	(0.0388)	
Municipality FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
Ν	13884	13884	13884	13884	13926

Table 5: Pre-trends in election results and income per capita - 2008-2013

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. The estimation sample includes municipalities for which electoral data in 2008 and 2013 are available. The placebo test considers outcome variables in 2008 and 2013. For observations in 2013, AS share is the fraction of asylum seekers in 2017 over total 2013 population at the municipality level, while it is equal to 0 for all municipalities in 2008.

5 Results

5.1 Political impact

Table 6 reports results from the estimation of the baseline model in Equation (1) for the share of votes by party. The arrival of asylum seekers significantly increases the support for antiimmigration parties, yet the impact is small in magnitude. A 1 p.p. change in the share of allocated asylum seekers, which is equivalent to 1.2 within-municipality standard deviations (s.d.= 0.82), is associated with an expected increase by 0.17 p.p. in the share of votes for anti-immigration parties in the election of the *Camera* and by 0.16 p.p. for the *Senato* election (Column 1). Considering that the within standard deviation of the share of votes for anti-immigration parties for the *Chamber of Deputies* election is equal to 9.6 p.p., the estimated effect is equivalent to $\frac{0.17}{9.6} = 1.7$ percent of within variation.

In Columns 2 and 3, the outcome is the share of votes for, respectively, *Lega* and FDI. The estimated coefficients show that the allocation of asylum seekers is positively correlated with the local share of preferences for both parties. Point estimates reveal that the impact is quantitatively higher for the share of votes for *Lega*, although it is not significantly different from 0 for the Senate election.

	1000		1 1 00 01 00 .	Babelline 1	0100010110		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	nti-immigratio	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Chan	nber of Deg	puties		
AS share	0.1742^{***}	0.1196^{**}	0.0579^{**}	-0.0027	-0.1116^{*}	-0.1265^{***}	0.0483
	(0.0572)	(0.0570)	(0.0246)	(0.0367)	(0.0579)	(0.0438)	(0.0321)
				Senate			
AS share	0.1649^{***}	0.0881	0.0717^{**}	0.0144	-0.1287^{**}	-0.1546^{***}	0.0505
	(0.0623)	(0.0558)	(0.0335)	(0.0492)	(0.0593)	(0.0552)	(0.0327)
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	13930	13930	13930	13930	13930	13930	13930

Table 6: Election results - Baseline regressions

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	A	nti-immigrati	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Chan	nber of Dep	puties		
AS share	0.2440***	0.1817^{***}	0.0640***	-0.0205	-0.2010***	-0.1022**	0.0413
	(0.0509)	(0.0508)	(0.0248)	(0.0367)	(0.0619)	(0.0427)	(0.0314)
				Senate			
AS share	0.2364^{***}	0.1471^{***}	0.0820**	-0.0047	-0.2162***	-0.1323**	0.0430
	(0.0528)	(0.0520)	(0.0327)	(0.0480)	(0.0601)	(0.0536)	(0.0320)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table 7: Election results - Municipality-level controls

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01.AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

The positive impact on the support for anti-immigration parties is offset by the loss in consensus for the center-left parties and M5S. An increase of one p.p. in the share of allocated asylum seekers leads to a reduction in the share of votes for center-left equal to 0.12 p.p. for the Chamber of Deputies election and to 0.15 p.p. for the Senate (Column 6). Likewise, the share of votes for M5S falls by 0.11 p.p. for the Chamber of Deputies election and by 0.12 for the Senate (Column 5). Lastly, the presence of asylum seekers does not seem to affect the share of preferences for PDL (Column 4) and the electoral turnout (Column 7).

Even though we leverage the dispersal policy design, we further test the robustness of our results to extra time-varying economic and demographic factors at the municipality level. For instance, at some point in time municipalities receiving asylum seekers might be simultaneously hit by negative economic shocks, experiencing a reduction in labor demand. This can lead to increased support for anti-immigration parties through channels other than direct exposure to asylum seekers. Similarly, population ageing may induce a "natural" shift in the share of votes for right-wing parties. In Table 7, we run the same regressions as above while controlling for municipality (log) income per capita and the share of the population over 65 years old (the latter is the only variable showing imbalance at the baseline). The introduction of this set of time-varying controls does not alter the direction and significance of the estimated impact, while the magnitude of point estimates is slightly higher. A 1 p.p. change in the share of asylum seekers increases the share of votes for anti-immigration parties by 0.24 p.p. in parliamentary elections. This corresponds to 2.5 percent of total withinvariation in vote shares.⁴⁵ In Appendix we report additional results while controlling for other potential confounding factors, which, however, are more endogenous and hence their uncritical inclusion may have some potential drawbacks. We report results while including the share of foreign born population at the municipality level (Table A2) and the share of asylum seekers hosted in SPRAR centers (Table A3). Results remain qualitatively and quantitatively unchanged. It is worth noting that the share of foreign born population may be directly affected by arrivals of asylum seekers (by their enrolment in the municipal registry offices)⁴⁶ and that the SPRAR system is directly influenced by political preferences at the municipality level.⁴⁷

In order to address any remaining possible concern related to some unobserved difference between municipalities that host asylum seekers and those that do not (something that we excluded from balance tests), we carry out two additional robustness checks. First, in order to address potential bias deriving from pre-existing political trends, in Table A6 in Appendix, we estimate the same

 $^{^{45}}$ In order to control for extra aggregate–level shocks that may play a role at the regional level, we run a more demanding specification with region–by–time fixed effects. Results are qualitatively the same (see Table A1 in Appendix.)

 $^{^{46}}$ In a different robustness check we further control for (log) native born population as well as the inverse of the latter. Results are unaffected (available upon request), which is reassuring also as for the potential 'share regression bias' (see Kronmal, 1993).

⁴⁷As described in the previous section, both asylum seekers hosted in SPRAR centers and municipalities that opened a SPRAR center represent a small fraction of the whole reception system. In particular, 700 municipalities opened a SPRAR center in 2014–2017, 438 of which hosted a CAS as well. In another robustness check we run our estimates excluding municipalities hosting a SPRAR center either at the beginning of the period or at any point in time during the period of 2014-2017, and results remain unchanged (see Tables A4 and A5 in Appendix).

model as in (1) adjusting for pre-trends in voting outcomes in the years before the refugee crisis. The vector of pre-trends variables essentially includes the change between 2008 and 2013 in electoral turnout and votes share for *Lega*, PDL and PD. Point estimates are not significantly affected. Finally, we estimate political impact as above while using a different (more balanced) estimation sample. Since we know when municipalities received asylum seekers up to 2019, we use the same former specification but including only those municipalities that received asylum seekers for the first time in 2018 and 2019 in the control group. In other words, we assess the political impact of asylum seekers (occurring in March 2018) while comparing those treated up to 2017 with those 'not treated yet' (i.e. to be treated in 2018–2019). Results are reported in Table A7 in Appendix and confirm the evidence provided above.

Furthermore, to assess our results are robust to the characteristics of the dispersal reception system at the local level, we dig deeper into the heterogeneity of both the spatial and time (duration) dimensions of the policy. Does the impact on voting vary with the length of exposure to asylum seekers? Is the shift of preferences towards anti-immigration parties driven by municipalities where the concentration of refugees is higher?

To answer the first question we test whether the impact of refugee allocation is different for municipalities where reception centers were opened for the first time in 2017 vs. those where CAS were opened before then, and for more than one year. Hence, we interact our regressor of interest as above with a dummy whether a CAS has been opened before 2017 (long term exposure). Results (reported in Table A8 in Appendix) point to positive support for anti-immigration parties being robust and greater (0.37 p.p.) in municipalities with a CAS that opened in the last year (recent exposure). Yet, the differential effect on anti-immigration voting is either negative or non-significant (especially by-party), so overall results point in the same direction (or are slightly attenuated) in those municipalities hosting a CAS center since 2016 or before.

We then exploit the granular variation in the Dispersal Policy, to investigate the role of the size and/or concentration of refugee reception centers. Indeed, given the high degree of heterogeneity in both the refugee reception system and municipality size in Italy, one concern is that our results may be driven by major hotspots (located in small areas) or large reception centers (hosting more than 100 people). We check this issue by excluding from our sample municipalities with (i) major governmental centers where first aid and identification procedures occur before asylum seekers are transferred to secondary reception centers; (ii) with centers hosting more than 100 people; (iii) municipalities with a high concentration of asylum seekers over the total resident population (i.e. in the top 1 percent of the share of asylum seekers). Results (reported in Appendix in Tables A9, A10 and A11 respectively) are in line with our baseline model and suggest that general findings on political support for anti–immigration parties are not driven by municipalities where either refugee centers are larger or the concentration of asylum seekers at the local level is higher.

5.2 Heterogeneity by municipality characteristics

In what follows, we allow the impact to differ according to municipality characteristics at the baseline. In particular, we test whether the effect is heterogeneous with respect to four local characteristics in 2013. First, we add to the model in Equation (1) the interaction between asylum seeker share and a dummy equal to 1 for municipalities with a population size above the 95th percentile in 2013.⁴⁸ Former studies find that the response to refugee allocation is different between urban and rural areas (e.g. Dustmann et al., 2019; Mayda, 2006). In municipalities at the top of the population distribution (mainly cities), the estimated impact on support for anti-immigration parties appears to be negative, while the opposite is true among low-populated areas. In Table 8 we report results on heterogeneous effects while focusing on the share of votes for anti-immigration parties only (Tables with estimates of heterogeneous effects on all political outcomes are reported in Table A12 to A14 in Appendix.) The coefficient of the interaction term in Column 1 shows that 1 p.p. increase in asylum seeker share in the largest municipalities is associated with a significant reduction in the share of votes for anti-immigration parties compared to smaller cities (the differential impact being 3.8. p.p. for the *Camera* and 3.5 p.p. for the *Senato* election). Point estimates for the effect among municipalities in the bottom 95 percent of the population distribution are not significantly different from the baseline regression in Table 6. Yet, the differential impact in antiimmigration voting across small and large cities is mainly driven by the electoral turnout, which is negatively affected by the share of asylum seekers in large municipalities compared to small ones (see Column (7) in Table A12 in Appendix.). In line with recent findings, these results confirm that the presence of asylum seekers has different effects on the political behavior of people in small vs. large communities, and indicate that it is crucial to understand the inherent characteristics that make a local area a suitable environment for the reception of asylum seekers.

Second, we study the heterogeneous effects with respect to the fraction of the foreign born population (i.e. regular immigrants) at the baseline. In particular, here we include the interaction between asylum seeker shares and a dummy equal to 1 for municipalities in the top 5 percent of the foreign population distribution in 2013.⁴⁹ With this test we aim to assess whether the level of

⁴⁸The 95th percentile of the municipality population distribution in 2013 is equal to 24,290 inhabitants.

⁴⁹The 95th percentile of the distribution across municipalities of the local share of immigrants in 2013 is equal

exposure to migrants in the pre-policy period induces different political reactions to newly arrived asylum seekers. Estimates in Column 2 reveal that a 1 p.p. increase in the asylum seeker share in municipalities with sizeable pre-treatment immigration is significantly associated with an increase in the share of votes for anti-immigration parties by more than 2 p.p. for both *Camera* and *Senato* elections, compared to municipalities with a lower immigration share. Interestingly, these findings do not seem to support the 'inter–group contact' hypothesis, according to which higher inter-ethnic contact may improve attitudes toward diversity. Our estimates suggest that pre–refugee–crisis exposure to migrants actually exacerbates the negative effect of asylum seekers on natives' attitude toward immigration. It is worth noting, though, that pre–crisis immigration may be rather different, in multiple dimensions (e.g. nationality, labor market integration, language proficiency etc.), to hosting asylum seekers.

Third, we assess whether the impact is heterogeneous with respect to a measure of pre-treatment local human capital by considering the interaction between asylum seeker share and a dummy equal to 1 for municipalities that are in the top 5 percent in terms of the educated share of the population (i.e. with a college degree or above) in 2011.⁵⁰ The presence of asylum seekers might elicit a different effect on local political preferences in municipalities with a higher fraction of high-skilled residents, since a smaller fraction of the population is exposed to the potential negative labor market externalities exerted by the arrival of new workers, generally at the bottom of earnings distribution.⁵¹ The results in Column 3 show that 1 p.p. increase in the share of asylum seekers in municipalities in the top 5 percent of the share of the highly-skilled population distribution is associated with a drop of around 0.5 p.p. in the share of votes for anti-immigration parties for both the *Camera* and *Senato*. Yet, in this case as well most of the heterogeneous effect seems to be explained by a fall in voter turnout (see Column (7) in Table A14 in Appendix.)

Finally, we add to the baseline model an interaction between the asylum seeker share and a dummy equal to 1 for municipalities with unemployment rate in 2011 above the 95th percentile⁵². In this way we test whether the response to asylum seeker inflows differs in case of more adverse economic conditions at the local level, as they reflect voters' economic anxiety and insecurity. Indeed, anti-immigration sentiments may be exacerbated if natives already hit by the consequences of the 2008

to 14.3 percent, where immigrants are defined as foreign born individuals.

⁵⁰The 95th percentile of the distribution across municipalities of the local share of population with college degree in 2011 is equal to 12.7 percent.

 $^{^{51}}$ Negative labor market externalities may arise from the fact that asylum seekers may take up employment opportunities or welfare services for local workers. As already mentioned though, this is unlikely to be the case in our setting. Indeed, if asylum seekers hosted in a CAS centers earn more than a certain amount, they are supposed to leave the center while waiting for the response about their eligibility for refugee status.

⁵²The 95th percentile of the unemployment rate distribution in 2011 is equal to 29.51 per cent.

economic crisis perceives the newcomers as an extra source of competition in the labour market. On the contrary, though, results in Column 4 show that 1 p.p. increase in asylum seeker share in municipalities with the highest unemployment rates is associated with a significant reduction in the share of votes for anti-immigration parties compared to the rest of the sample. Results in A15 in Appendix indicate that the loss of consensus for anti-immigration parties is offset by a gain in the electoral support for *Movimento 5 stelle*, which is more in line with the economic roots of pure populistic political responses. These findings suggest that the far–right anti-immigration backlash is not triggered by adverse economic circumstances. We will explore this direction deeper in the next section.

	(1)	(2)	(3)	(4)
	Anti-immigration	Anti-immigration	Anti-immigration	Anti-immigration
		Chamber of		
AS share	0.2460^{***}	0.2121^{***}	0.2609***	0.3173***
	(0.0509)	(0.0517)	(0.0520)	(0.0517)
top 5% largest municipalities \times AS share	-3.8414^{***}			
	(0.9150)			
top 5% share of for eigners \times AS share		2.3489^{***}		
		(0.3553)		
top 5% share of uni degree \times AS share			-0.5288^{***}	
			(0.1510)	
top 5% unemployment rate \times AS share				-1.4215^{***}
				(0.4343)
		Ser	nate	
AS share	0.2382^{***}	0.2088^{***}	0.2537^{***}	0.3044^{***}
	(0.0526)	(0.0549)	(0.0526)	(0.0497)
top 5% largest municipalities \times AS share	-3.5018^{***}			
	(0.8848)			
top 5% share of for eigners \times AS share		2.0312^{***}		
		(0.3731)		
top 5% share of uni degree \times AS share			-0.5419^{***}	
			(0.1869)	
top 5% unemployment rate \times AS share				-1.3180***
				(0.4020)
Municipality controls	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930

Table 8: Election results - Interaction with municipality characteristics at the baseline

6 Mechanisms

6.1 The Economic Effects of Refugee Reception

Refugee migration can affect political outcomes through economic and non-economic channels. In particular, vote shares for anti-immigration parties may increase if Italian citizens bear a net cost rather than a benefit from hosting asylum seekers in their cities (Ortega, 2005; Mayda, 2006). In this section, we assess whether the share of refugees allocated by the dispersal policy changed economic outcomes and prosperity in receiving municipalities. By employing the same fixed effects model as in Equation (1) above, we estimate the impact of refugee allocation on local income, welfare expenditure, natives' migration flows and population size at the municipality level.

We report results on the economic effects in Table 9, where we consider $t_1 = 2017$ our endline (i.e. before the 2018 national election). Column 1 reports the impact of the share of allocated asylum seekers on municipality (log) income per capita (of both natives and immigrants).⁵³ The estimated coefficient is negative and significant, with a one p.p. increase in asylum seeker share being associated with a 0.31 percent reduction in income per capita. This seems to suggest that asylum seekers may take a toll on the local economy, for instance, by displacing local workers out of employment or depressing local wages. However, considering the limited labor market opportunities for refugees hosted in CAS, this is unlikely to be a relevant mechanism.⁵⁴ In fact, the drop in income per capita may be mechanically induced by the arrival of new residents (i.e. asylum seekers themselves, when registered among residents). These inflows inflate the denominator of income per capita, while bringing little or no contribution to the numerator if new residents are non-income earning asylum seekers. We investigate this mechanical channel in Columns 2 and 3, where our dependent variables are the numerator (log income) and the denominator (log population) respectively. We find that a 1 p.p. increase in asylum seekers' share is significantly correlated with a decline of 0.1 percent in total income (Column 2) at the local level, and an increase of 0.21 percent in population size (Column 3).

Asylum seekers thus seem to bring a positive net contribution to population growth, while being correlated with a decline in total income. If the latter is not related to the labor market channel,

⁵³Our data source does not allow to distinguish natives' income from income of immigrants and/or refugees.

⁵⁴As already mentioned, less than 10 per cent of asylum seekers hosted in the Italian reception system had a job contract between 2013 and 2017, thus competition between asylum seekers and local workers in the labor market is unlikely to be a relevant channel in our setting.

it may be the result of population dynamics involving asylum seekers replacing natives, or former migrants, in the local population. While using natives and the foreign borns as dependent variables, Column 4 and 5 show that the share of asylum seekers is negatively related to native population growth and positively correlated with the growth in the number of foreign born residents. The latter is a mechanical effect while the former may be linked to the geographic 'displacement' of natives by asylum seekers, which may explain the total income drop (since, unlike natives, asylum seekers earn little or no income).⁵⁵

We test this by checking whether the allocation of asylum seekers is associated with the crosssectional variation in native population dynamics (internal migration, mortality and natality) over the time span considered. We show results in Table 10, where in Columns 1–3 outcome variables are population flows (between January 1st, 2013 and January 1st, 2018) as a fraction of 2013 municipality population.⁵⁶ Results show no significant geographical mobility or sorting across municipalities by natives in response to the inflow of asylum seekers.

Displacement of natives may likewise be related to (intergenerational) replacement. Indeed, from the balance test in Table 4 we observe a positive correlation between the share of elderly population and the allocation of asylum seekers at the local level. Municipalities more exposed to the dispersal policy are also those with the highest share of elderly population at t_0 , and are likely to experience higher mortality rates, lower birth rates and lower native population growth in the following years. The different age-structure at the baseline may therefore be responsible for the negative correlation between asylum seekers and native population growth. We test this hypothesis in Columns 4–7 in Table 10, where we estimate the cross-sectional correlation between asylum seeker allocation and death and birth rates between 2013 and 2018 over total population in 2013. We find that the share of asylum seekers is indeed significantly and positively correlated with the death rate during the treatment period, both among the total and native population, while it is negatively associated with the birth rate.

To account for the heterogeneity in these demographic dynamics, we estimate again the impact on the (log) native population, adjusting for the death and birth rates interacted with a year (2017) dummy (Column 6 in Table 9). The estimated effect is not significant, with a point estimate

⁵⁵Geographical displacement can be driven by negative labor market externalities, i.e. asylum seekers reducing employment opportunities and compensations for local workers, as well as motivated by asylum seekers entering residents' utility function as a local dis-amenity in the framework of residential location choice.

 $^{^{56}}$ We use ISTAT data on new resident new registrations and cancellations, from and toward other municipalities (or abroad), respectively, to construct measures of inflows and outflows. Their difference yields the net migration flow, which we standardize with the population in 2013. We use cross-sectional specifications here in order to compute all yearly population changes, which may overlooked by using a panel specification over 5-years span.

equal to zero. Hence, municipalities receiving more asylum seekers are experiencing lower native population growth through a higher number of deaths and lower number of births. Once we control for these potential confounding factors, the negative and significant correlation between asylum seekers share and native population growth vanishes. We finally re-estimate the effect of asylum seekers on (log) total income with the inclusion of the share of deaths as a control (Column 7 in Table 9). This allows us to check whether a higher mortality rate among municipalities more treated by the dispersal policy, explains, at least partially, the negative impact on total income growth. Results show that, after conditioning for mortality rate, there is no significant correlation between asylum seeker share and total income growth.⁵⁷

Overall, our results point to small-to-zero income effects, driven by the allocation of asylum seekers in municipalities with a high share of elderly population, who are eventually replaced. In other words, according to the dispersal policy scheme, municipalities hosting refugees (low or no-income earners) may also experience higher mortality rates (of income earners), and this may explain the total income effect.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	log(income p.c.)	log(income)	log(pop.)	log(native pop.)	log(foreign pop.)	log(native pop.)	$\log(\text{income})$
AS share	-0.0031***	-0.0010**	0.0021***	-0.0014***	0.0532^{***}	-0.0000	-0.0002
	(0.0008)	(0.0004)	(0.0007)	(0.0004)	(0.0138)	(0.0003)	(0.0005)
death rate (2013-2017) \times Time						-0.0092***	-0.0106***
						(0.0006)	(0.0007)
birth rate (2013-2017) \times Time						0.0183***	
· · · · · ·						(0.0006)	
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table 9: Effects on municipality income and population

¹ Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Death rate (2013-2017) is the municipality fraction of Italian citizens who died between 2013 and 2017 over population in 2013. Birth rate (2013-2017) is the municipality fraction of Italian citizens born between 2013 and 2017 over population in 2013. Both are interacted with a year dummy in order to compare municipalities with similar levels of the two variables in the treatment period (2013-2017).

While we find weak evidence of income effects at the local level, we next test average welfare effects in the allocation of municipality public expenditure. Indeed, the latter may be diluted or

⁵⁷We repeat the same analysis at a slightly different unit, which is the 'labor Market Area' (LMA) instead of the municipality. LMAs are sub-regional geographical areas in Italy where the bulk of the labor force lives and works, and where establishments can find the largest amount of the labor force necessary to occupy the offered jobs. This allows us to check whether economic effects are activated by the treatment at a level that is more economically relevant than the one defined by municipalities' boundaries. Results are reported in Table A16 in Appendix and confirm our main results on the economic impact at the municipality level.

	1	5			0 ,	.,	57
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Natives' migration		Death	n rate	Birth	ı rate	
	Net flows	Inflows	Outflows	Total	Natives	Total	Natives
AS share	-0.0104	0.0410	0.0514	0.0705**	0.0693**	-0.0367***	-0.0403***
	(0.0265)	(0.0452)	(0.0332)	(0.0282)	(0.0277)	(0.0077)	(0.0082)
Ν	6965	6965	6965	6965	6965	6965	6965

Table 10: Population dynamics (natives' internal migration, mortality, natality)

Notes: Standard errors in parentheses clustered at province level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the share of asylum seekers allocated to the municipality in 2017 on 2013's total population (average across different measurements when available). The dependent variable in specifications in Col.1—3 considers the share of people who moved across municipalities, between January 1st 2013 and January 1st 2018, over the municipality population in 2013. The dependent variable in Column 2, for example, represents the number of natives who moved into the municipality in the above time span, standardized over population in 2013. The dependent variables in Columns 4 and 5 considers the share of people who died, between January 1st 2018 and January 1st 2018 over the municipality population in 2013. The dependent variables in Columns 6 and 7 represent the share of birth, between January 1st 2013 and January 1st 2013.

may decrease as soon the pool of users grows, as in the case of substantial asylum seeker inflows. The increase in competition for public resources may also be the reason behind the rising support for anti-immigration parties. In Table 11, we regress the (log) municipality per–user expenditure for public services in various categories on the share of allocated asylum seekers ⁵⁸. If we exclude a significant increase in the expenditure for people with severe addictions, the share of asylum seekers is not significantly correlated with the expenditure in any of the categories considered. There is therefore no evidence that the arrival of asylum seekers leads to a reduction in the availability of public resources for local residents.

				1 1 1			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Log(Total)	Log(Families)	Log(Addicted)	Log(Poverty)	Log(Elderly)	Log(Disabilities)	Log(Immigrants)
AS share	0.0006	-0.0013	0.0062^{***}	0.0045	0.0014	-0.0078	0.0125
	(0.0063)	(0.0074)	(0.0021)	(0.0072)	(0.0100)	(0.0305)	(0.0111)
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	13604	13604	13604	13604	13604	13604	13604

Table 11: Municipality expenditure

Notes: Standard errors in parentheses clustered at the municipality level. The estimation sample includes municipalities for which data on expenditure for public services are available. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level.

⁵⁸We consider the dependent variables and the share of asylum seekers until 2016 as ISTAT data on municipality expenditure are not available after that year. In Table A17 in Appendix we report the same estimation model while including additional controls (log pc income and the share of population over 65), and results are confirmed.

6.2 The Role of Political Propaganda

Since electoral preferences do not seem to be rooted in actual economic mechanisms, in this section we examine whether voting behavior may be oriented by ideological or political drivers. In particular, by exploiting the fact that refugees are hosted in Italy during electoral campaign, we assess to what extent being exposed to political propaganda by right-wing candidates (including *Lega*'s Leader Salvini and FDI's leader Meloni) can exacerbate the political impact of refugee reception. We estimate the same fixed effects model as in Equation (1) above, adding the interaction between the share of asylum seekers and a dummy equal to 1 if a rally or political event organized by the main right-wing parties (Lega, FDI, PDL) took place in the municipality during the electoral campaign. We include an indicator for anti-immigration rallies all together, and in different specifications we include party-specific rally indicators separately. We focus on right-wing propaganda delivered in person by political candidates, hence over and above political stances fuelled by social media and engagement (the time-invariant component of which is controlled for by municipality fixed effects). This is in line with major relevance of traditional ground campaigning and with the literature about the role that charismatic leaders often assume to drive the electoral success of populist radical right parties.

A total of 851 municipalities were visited by right-wing candidates between January 1st and March, 4th, the date of the general elections (Figure A3 in Appendix). Overall, right-wing campaigns were fairly balanced across municipalities according to the share of hosted refugees (about half of the municipalities with at least a right-wing rally also hosted refugees in a CAS). Results in Table 12 show that right-wing rallies have a significant additional effect on the share of votes for antiimmigration parties. While the political impact of the share of refugees at the municipality level is in line with our baseline model (slightly smaller), the coefficients of the interaction terms are significantly positive and greater in magnitude, suggesting that the anti-immigration backlash for refugee hosting is significantly higher in municipalities visited by right-wing electoral rallies. Results in Column 2 indicate that this effect is mostly driven by *Lega*.

	(1)	(2)	(3)	(4)
	A	nti-immigrati	on	PDL
	Total	Lega Nord	FDI	
		Chamb	er of Deput	ies
AS share	0.2282***	0.1662^{***}	0.0640***	-0.0168
	(0.0502)	(0.0507)	(0.0248)	(0.0364)
anti-immigration campaign \times AS share	1.1178^{**}			
	(0.5108)			
campaign Lega \times AS share		1.5692^{***}		
		(0.2775)		
campaign FDI \times AS share			0.0035	
			(0.3992)	
campaign PDL other \times AS share				-1.3794^{*}
				(0.7850)
			Senate	
share AS	0.2232***	0.1334^{***}	0.0839***	-0.0023
	(0.0533)	(0.0516)	(0.0325)	(0.0480)
anti-immigration campaign \times AS share	0.9342^{**}			
	(0.4519)			
campaign Lega \times AS share		1.3964^{***}		
		(0.3035)		
campaign FDI \times AS share			-0.4164	
			(0.5019)	
campaign PDL \times AS share				-0.8820
				(0.7358)
Municipality controls	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930

Table 12: Election results - Interaction with political propaganda

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

To better measure the 'dog-whistle' effect of being directly exposed to xenophobic propaganda, among electoral campaigns we identify those events held by hate-speech candidates. To do so we match the Amnesty International "hate-speech" dataset with our data tracking the geo-localized electoral campaigns of right-wing candidates. We identify 36 "hate-speech" candidates who also held a rally before 2018 elections. In Table 13 right-wing propaganda exposure is measured by a dummy equal to 1 if a "hate-speech" candidate held an electoral rally in municipality m between January and March 2018. The dummy is equal to 1 for 231 municipalities. As before, the coefficients of the interaction terms are significantly positive and even larger in magnitude, suggesting that rallies by hate-speech candidates have a greater effect on the share of votes for anti–immigration parties than generic right-wing rallies.

	(1)	(2)	(3)	(4)		
	Anti-immigration	Lega	FDI	PDL		
	Camera dei Deputati					
AS share	0.2408^{***}	0.1793^{***}	0.0632**	-0.0198		
	(0.0510)	(0.0508)	(0.0248)	(0.0367)		
rally by hate-speech cand. \times AS share	2.6404^{***}	1.9357^{***}	0.6711^{***}	-0.5652		
	(0.8974)	(0.7467)	(0.2296)	(0.4889)		
		Senate)			
AS share	0.2331^{***}	0.1448^{***}	0.0811**	-0.0033		
	(0.0530)	(0.0519)	(0.0329)	(0.0480)		
rally by hate-speech cand. \times AS share	2.7210^{**}	1.9570^{**}	0.7192^{***}	-1.0916^{*}		
	(1.0704)	(0.9142)	(0.2415)	(0.5778)		
Municipality controls	Yes	Yes	Yes	Yes		
Municipality FE	Yes	Yes	Yes	Yes		
Time FE	Yes	Yes	Yes	Yes		
Ν	13930	13930	13930	13930		

Table 13: Election results - overall right-wing propaganda exposure – with controls

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

Overall, these findings provide evidence in support of the role of anti-immigration political propaganda in boosting divisive negative views towards alien groups through a 'dog-whistle effect', hence mobilizing voter support and boosting their electoral prominence. The effect is greater the more xenophobic views are made explicit by the member of anti-immigration parties, and leaves venue for accurate information campaigns and active programs against nativist and divisive stances.

7 Conclusions

This paper contributes to a growing literature about the economic and political impact of asylum seekers on receiving societies by exploiting an unprecedented surge in the arrival of asylum seekers and informal immigrants to Italy, mainly through the Mediterranean route. The Italian Dispersal Policy, implemented from 2014 onwards, allocated these massive inflows into CAS reception centers

across Italian municipalities on a quasi-random basis. This policy was in line with what was implemented in the rest of the EU region, with the aim of spreading the 'burden' of refugee reception, increase the perception of fairness among the local population, prompt asylum seeker integration and avoid native public discontent.

We use unique administrative data on the number of asylum seekers allocated to each Italian municipality between 2014 and 2017 to study the effect of the dispersal policy on the local share of votes for anti-immigration parties, as well as parties over the rest of the political spectrum, between two national elections held in 2013 and 2018.

After showing that the allocated share of asylum seekers is fairly balanced with respect to local characteristics at the baseline, we find that a 1 p.p. increase in the refugee share in 2017, at the peak of the refugee crisis, is correlated with a small but significant increase in the vote share for anti-immigration parties, leading to a drop in the support for both the main populist party and the center-left. This effect is robust to the inclusion of a set of time-varying controls, in particular municipality income per capita and local demographic composition, and it is not associated to any significant change in voter turnout.

We also test whether the effect is heterogeneous with respect to a group of municipality characteristics in the pre-treatment period. We detect, similarly to Dustmann et al. (2020), opposite effects when we distinguish between urban and rural areas, with asylum seekers actually reducing the support for anti-immigration parties in the largest municipalities. Yet, this effect is mainly associated with a drop in voter turnout in large cities hosting asylum seekers. Moreover, different from the predictions of contact-theory, our findings show that higher exposure to immigration in the pre-treatment period exacerbates the negative effect of the presence of asylum seekers on natives' attitudes toward migrants and increases support for anti-immigration parties. We also find that in municipalities with higher share of residents with at least a college degree, the presence of asylum seekers reduces the share of votes for anti-immigration parties, along with electoral turnout.

We explore whether refugee migration affected political outcomes through economic channels. We examine the impact of asylum seekers first on income per capita and then on per user expenditure for local public services. Both analyses do not yield significant short-run effects on actual economic costs, supporting the idea that ideological traits or culture-related fears may have driven voting behavior and political outcomes in Italy during the refugee crisis. Thus, we investigate the role of political propaganda during the electoral campaign, in mobilizing voters' support. Our results support the idea that radical-right propaganda boosts divisive negative views about the presence of asylum seekers. Indeed, the anti–immigration backlash for refugee hosting is higher in municipalities where right–wing propaganda took place. The effect is larger where anti-immigration propaganda, measured by hate speech, is harsher.

These findings offer novel insights to the policy debate about the modalities in which dispersal policies and integration models for asylum seekers and refugees should be implemented. Focusing on the ordinary reception model in the same Italian context, Gamalerio et al. (2020) find that hosting refugees in a reception system managed by local administrations and integrated in the local context can stem extreme-right support. Yet, the latter model proved to be insufficient and the vast majority of local administrations in Italy rejected the possibility of managing integrated refugee hosting. Overall, the anti-immigration backlash observed in our study indicates that a model of refugee hosting based on 'temporary' reception centers—without the provision of adequate integration services, the involvement of local institutions in their management and also adequate information campaigns—can generate discontent and be politically costly.

Overall, asylum policies dominated by short-term strategies are not likely to be responsive to realities on the ground. Efforts to craft a global consensus on migration are too often hampered by demands for quick solutions. More integrated policies (e.g. between the central government and local administrations) and long-term reception measures are better suited to address local needs for the benefit of newcomers and the communities that receive them.

References

References

ActionAid and Openpolis (2018). Centri d'Italia.

- Albrecht, S., Ghidoni, R., Cettolin, E., and Suetens, S. (2020). Exposure to ethnic minorities changes attitudes to them. University of Milan Bicocca Department of Economics, Management and Statistics Working Paper, (449).
- Alesina, A. F. and Tabellini, M. (2020). The political effects of immigration: Culture or economics?
- Allport, G. W., Clark, K., and Pettigrew, T. (1954). The nature of prejudice.
- Altındağ, O. and Kaushal, N. (2020). Do refugees impact voting behavior in the host country? evidence from Syrian refugee inflows to Turkey. *Public Choice*, pages 1–30.
- Baez, J. E. (2011). Civil wars beyond their borders: The human capital and health consequences of hosting refugees. *Journal of development economics*, 96(2):391–408.
- Bansak, K., Hainmueller, J., and Hangartner, D. (2016). How economic, humanitarian, and religious concerns shape European attitudes toward asylum seekers. *Science*, 354(6309):217–222.
- Barone, G., D'Ignazio, A., de Blasio, G., and Naticchioni, P. (2016). Mr. Rossi, Mr. Hu and politics. the role of immigration in shaping natives' voting behavior. *Journal of Public Economics*, 136:1– 13.
- Blanchflower, D. G. and Shadforth, C. (2009). Fear, unemployment and migration. *The Economic Journal*, 119(535):F136–F182.
- Borjas, G. J. (2017). The labor supply of undocumented immigrants. *Labour Economics*, 46:1–13.
- Boustan, L. P. (2010). Was postwar suburbanization "white flight"? evidence from the black migration. *The Quarterly Journal of Economics*, 125(1):417–443.
- Bratti, M., Deiana, C., Havari, E., Mazzarella, G., and Meroni, E. C. (2020). Geographical proximity to refugee reception centres and voting. *Journal of Urban Economics*, 120:103290.
- Bursztyn, L., Egorov, G., Enikolopov, R., and Petrova, M. (2019). Social media and xenophobia: evidence from Russia. Technical report, National Bureau of Economic Research.
- Card, D. (1990). The impact of the Mariel boatlift on the Miami labor market. *ILR Review*, 43(2):245–257.
- Card, D., Dustmann, C., and Preston, I. (2012). Immigration, wages, and compositional amenities. Journal of the European Economic Association, 10(1):78–119.

- Chamber of Deputies (2017). Relazione sul sistema di protezione e di accoglienza dei richiedenti asilo. commissione parlamentare di inchiesta.
- Clemens, M., Huang, C., and Graham, J. (2018). The economic and fiscal effects of granting refugees formal labor market access. Technical report, Center for Global Development.
- Clemens, M. A. and Hunt, J. (2019). The labor market effects of refugee waves: reconciling conflicting results. *ILR Review*, 72(4):818–857.
- Coates, T.-N. (2017). The first white president: The foundation of Donald Trump's presidency is the negation of Barack Obama's legacy. *The Atlantic*, 320(5):6.
- Dröes, M., Koster, H. R., et al. (2020). A world divided: Refugee centers, house prices, and household preferences. Technical report, CEPR Discussion Papers.
- Dustmann, C., Schönberg, U., and Stuhler, J. (2016). The impact of immigration: Why do studies reach such different results? *Journal of Economic Perspectives*, 30(4):31–56.
- Dustmann, C., Vasiljeva, K., and Piil Damm, A. (2019). Refugee migration and electoral outcomes. The Review of Economic Studies, 86(5):2035–2091.
- Edo, A., Giesing, Y., Öztunc, J., and Poutvaara, P. (2019). Immigration and electoral support for the far-left and the far-right. *European Economic Review*, 115:99–143.
- Entorf, H. and Lange, M. (2019). Refugees welcome? understanding the regional heterogeneity of anti-foreigner hate crimes in germany. Technical report, ZEW Discussion Papers.
- Eurostat (2020). Asylum and managed migration database, eu 27, 2008-2019.
- Facchini, G., Mayda, A. M., and Mendola, M. (2013). What drives individual attitudes towards immigration in South Africa? *Review of International Economics*, 21(2):326–341.
- Fletcher, R., Newman, N., and Schulz, A. (2020). A mile wide, an inch deep: Online news and media use in the 2019 UK general election. *Available at SSRN 3582441*.
- Foged, M. and Peri, G. (2016). Immigrants' effect on native workers: New analysis on longitudinal data. American Economic Journal: Applied Economics, 8(2):1–34.
- Freddi, E. (2019). Do people avoid morally relevant information? evidence from the refugee crisis. *Review of Economics and Statistics*, pages 1–45.
- Friedberg, R. M. (2001). The impact of mass migration on the Israeli labor market. The Quarterly Journal of Economics, 116(4):1373–1408.
- Gamalerio, M. (2018). Not welcome anymore: the effect of electoral incentives on the reception of refugees.

- Gamalerio, M., Luca, M., Romarri, A., and Viskanic, M. (2020). Is this the real life or just fantasy? refugee reception, extreme-right voting, and broadband internet. *Refugee reception*, extreme-right voting, and broadband internet (August 14, 2020).
- Gehrsitz, M. and Ungerer, M. (2017). Jobs, crime, and votes: A short-run evaluation of the refugee crisis in Germany.
- Gennaioli, N., Tabellini, G., et al. (2019). Identity, beliefs, and political conflict. Technical report, CESifo.
- Gerdes, C. and Wadensjö, E. (2008). The impact of immigration on election outcomes in Danish municipalities. Technical report, IZA Discussion Papers.
- Gessler, T., Tóth, G., and Wachs, J. (2019). No country for asylum seekers? how short-term exposure to refugees influences attitudes and voting behavior in Hungary.
- Glitz, A. (2012). The labor market impact of immigration: A quasi-experiment exploiting immigrant location rules in Germany. *Journal of Labor Economics*, 30(1):175–213.
- Goodin, R. E. (2008). Innovating democracy: Democratic theory and practice after the deliberative turn. OUP Oxford.
- Grosjean, P. A., Masera, F., and Yousaf, H. (2020). Whistle the racist dogs: Political campaigns and police stops. *Available at SSRN*.
- Guriev, S. and Papaioannou, E. (2020). The political economy of populism. Available at SSRN 3542052.
- Hainmueller, J. and Hopkins, D. J. (2014). Public attitudes toward immigration. Annual review of political science, 17:225–249.
- Halla, M., Wagner, A. F., and Zweimüller, J. (2017). Immigration and voting for the far right. *Journal of the European Economic Association*, 15(6):1341–1385.
- Hangartner, D., Dinas, E., Marbach, M., Matakos, K., and Xefteris, D. (2019). Does exposure to the refugee crisis make natives more hostile? *American Political Science Review*, 113(2):442–455.
- Hatton, T. J. (2020). Asylum migration to the developed world: Persecution, incentives, and policy. *Journal of Economic Perspectives*, 34(1):75–93.
- Hunt, J. (1992). The impact of the 1962 repatriates from algeria on the French labor market. *ILR Review*, 45(3):556–572.
- Kronmal, R. A. (1993). Spurious correlation and the fallacy of the ratio standard revisited. *Journal* of the Royal Statistical Society: Series A (Statistics in Society), 156(3):379–392.
- Levi, E., Mariani, R. D., and Patriarca, F. (2020). Hate at first sight? dynamic aspects of the electoral impact of migration: the case of Ukip. *Journal of Population Economics*, 33(1):1–32.

- Lonsky, J. (2021). Does immigration decrease far-right popularity? evidence from Finnish municipalities. *Journal of Population Economics*, 34(1):97–139.
- López, I. H. (2015). Dog whistle politics: How coded racial appeals have reinvented racism and wrecked the middle class. Oxford University Press.
- Mayda, A. M. (2006). Who is against immigration? a cross-country investigation of individual attitudes toward immigrants. *The review of Economics and Statistics*, 88(3):510–530.
- Mayda, A. M., Peri, G., and Steingress, W. (2018). The political impact of immigration: Evidence from the United States. Technical report, National Bureau of Economic Research.
- Maystadt, J.-F. and Verwimp, P. (2014). Winners and losers among a refugee-hosting population. Economic development and cultural change, 62(4):769–809.
- Mendez, I. and Cutillas, I. M. (2014). Has immigration affected Spanish presidential elections results? *Journal of Population Economics*, 27(1):135–171.
- Müller, K. and Schwarz, C. (2018). Fanning the flames of hate: Social media and hate crime. Journal of the European Economic Association.
- Ortega, F. (2005). Immigration quotas and skill upgrading. *Journal of Public Economics*, 89(9-10):1841–1863.
- Ottaviano, G. I. and Peri, G. (2012). Rethinking the effect of immigration on wages. *Journal of the European economic association*, 10(1):152–197.
- Otto, A. H. and Steinhardt, M. F. (2014). Immigration and election outcomes—evidence from city districts in Hamburg. *Regional Science and Urban Economics*, 45:67–79.
- Peri, G. and Yasenov, V. (2019). The labor market effects of a refugee wave synthetic control method meets the Mariel boatlift. *Journal of Human Resources*, 54(2):267–309.
- Robinson, V., Anderson, R., and Musterd, S. (2004). Spreading the burden? a review of policies to disperse asylum seekers and refugees. *Refugee Survey Quarterly*, 23(3).
- Rodrik, D. (2020). Why does globalization fuel populism? economics, culture, and the rise of right-wing populism. Technical report, National Bureau of Economic Research.
- Romarri, A. (2020). Does the internet change attitudes towards immigrants? evidence from Spain. Technical report, Institut d'Economia de Barcelona (IEB).
- Rozo, S. and Vargas, J. F. (2020). Brothers or invaders? how crisis-driven migrants shape voting behavior. *Journal of Development Economics, Forthcoming.*
- Steinmayr, A. (2020). Contact versus exposure: Refugee presence and voting for the far-right. *Review of Economics and Statistics*, pages 1–47.

UNHCR (2018). Mediterranean situation - UNHCR data portal.

Volkens, A., Burst, T., et al. (2020). The Manifesto data collection. Manifesto Project (MRG/CMP/MARPOR). version 2020b.

A Appendix



Figure A1: Asylum Seekers in CAS and SPRAR Systems, 2014–2019

Figure A2: Vote shares for anti-immigration parties (Lega + FDI), 2013–2018



Figure A3: Geocoded right-wing rallies/ campaign events btw Jan-March 2018 (via Twitter data)



Table A1: Election results - Region by time fixed effects

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Aı	Anti-immigration		PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Cham	ber of Dep	uties		
AS share	0.1227^{***}	0.0624^{*}	0.0666***	0.0128	-0.0262	-0.1337***	0.0350
	(0.0390)	(0.0351)	(0.0236)	(0.0336)	(0.0361)	(0.0345)	(0.0293)
				Senate			
AS share	0.1421***	0.0450	0.0923***	0.0114	-0.0440	-0.1195***	0.0383
	(0.0423)	(0.0347)	(0.0296)	(0.0417)	(0.0300)	(0.0368)	(0.0299)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region-Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	13930	13930	13930	13930	13930	13930	13930

				Ű			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	nti-immigratio	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Char	nber of De	puties		
AS share	0.3981^{***}	0.3332^{***}	0.0603**	-0.0307	-0.3950***	-0.0334	0.0281
	(0.0643)	(0.0617)	(0.0244)	(0.0383)	(0.0971)	(0.0435)	(0.0309)
				Senate			
AS share	0.3992^{***}	0.3033***	0.0828^{**}	-0.0267	-0.4101***	-0.0309	0.0313
	(0.0562)	(0.0684)	(0.0322)	(0.0474)	(0.0904)	(0.0516)	(0.0314)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	13930	13930	13930	13930	13930	13930	13930

Table A2: Election results - Control for foreign-born population

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01.AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	An	ti-immigratio	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Chan	nber of Dep	outies		
AS share	0.2494^{***}	0.1865^{***}	0.0645^{***}	-0.0215	-0.2074***	-0.1015**	0.0411
	(0.0508)	(0.0508)	(0.0246)	(0.0368)	(0.0626)	(0.0427)	(0.0314)
AS share in SPRAR	-0.9558^{***}	-0.8521^{***}	-0.0799	0.1847	1.1243^{**}	-0.1194	0.0254
	(0.3551)	(0.3049)	(0.1668)	(0.2025)	(0.4493)	(0.1541)	(0.1349)
				Senate			
AS share	0.2421^{***}	0.1516^{***}	0.0832**	-0.0062	-0.2231***	-0.1309**	0.0429
	(0.0523)	(0.0523)	(0.0325)	(0.0480)	(0.0606)	(0.0536)	(0.0320)
AS share in SPRAR	-1.0052^{***}	-0.7835^{***}	-0.2090	0.2687	1.2052^{***}	-0.2483	0.0228
	(0.3528)	(0.2996)	(0.1425)	(0.1789)	(0.4398)	(0.1991)	(0.1357)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table A3: Election results - Control for AS share hosted in SPRAR projects

		0	1	0				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
	A	nti-immigrati	on	PDL	M5S	center-left	Turnout	
	Total	Lega Nord	FDI			PD + Left		
		Cha			ber of Deputies			
AS share	0.2303***	0.1665^{***}	0.0643^{**}	-0.0238	-0.1909***	-0.1119^{***}	0.0369	
	(0.0508)	(0.0512)	(0.0250)	(0.0368)	(0.0605)	(0.0433)	(0.0313)	
				Senate				
AS share	0.2256^{***}	0.1322^{**}	0.0858^{***}	-0.0062	-0.2028***	-0.1410***	0.0393	
	(0.0532)	(0.0521)	(0.0327)	(0.0482)	(0.0589)	(0.0547)	(0.0319)	
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Ν	13294	13294	13294	13294	13294	13294	13294	

Table A4: Election results excluding municipalities hosting a SPRAR before treatment

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

			0	-p	C		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	A	Anti-immigration			M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Chan	nber of Dep	puties		
AS share	0.2242***	0.1741^{***}	0.0489***	-0.0372	-0.1821***	-0.0970**	0.0225
	(0.0534)	(0.0505)	(0.0179)	(0.0384)	(0.0647)	(0.0402)	(0.0320)
				Senate			
AS share	0.2301***	0.1332^{**}	0.0868**	-0.0442	-0.2068***	-0.1196**	0.0233
	(0.0550)	(0.0540)	(0.0351)	(0.0378)	(0.0618)	(0.0527)	(0.0319)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	12516	12516	12516	12516	12516	12516	12516

Table A5: Election results excluding municipalities ever hosting a SPRAR

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	Anti-immigration			M5S	$\operatorname{center-Left}$	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Char	nber of De	eputies		
AS share	0.1910^{***}	0.1461^{***}	0.0516^{**}	-0.0067	-0.1512^{***}	-0.1170^{***}	0.0238
	(0.0417)	(0.0396)	(0.0246)	(0.0359)	(0.0546)	(0.0396)	(0.0266)
				Senate			
AS share	0.1770^{***}	0.1046^{**}	0.0686^{**}	0.0014	-0.1471^{***}	-0.1526^{***}	0.0213
	(0.0458)	(0.0458)	(0.0309)	(0.0467)	(0.0527)	(0.0526)	(0.0267)
Pre-trends voting $(2008-13) \times \text{Time}$	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13884	13884	13884	13884	13884	13884	13884

Table A6: Election results -	control for	pre-trends in	voting outcomes
------------------------------	-------------	---------------	-----------------

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. The estimation sample includes municipalities for which data on 2008 and 2013 national elections are available. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population. The vector of pre-trends includes the change in turnout and votes share for each party competing in 2008 and 2013 (*Lega*, PDL and PD) interacted with a year dummy.

Table A7: Election results – estimation sample with 'yet–to–be–treated' municipalities as control group

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	A	nti-immigrati	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Char	nber of De	puties		
AS share	0.1376^{**}	0.0873	0.0485^{*}	0.0550	0.0075	-0.1258^{***}	0.1382^{***}
	(0.0574)	(0.0551)	(0.0255)	(0.0365)	(0.0647)	(0.0447)	(0.0444)
				Senate			
AS share	0.1226^{*}	0.0519	0.0639^{*}	0.0736	-0.0076	-0.1304^{**}	0.1437^{***}
	(0.0668)	(0.0542)	(0.0370)	(0.0526)	(0.0675)	(0.0562)	(0.0460)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	5950	5950	5950	5950	5950	5950	5950

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. The estimation sample includes municipalities that received asylum seekers between 2014 and 2019. Since our period of observation for the political impact is 2013–2018, those municipalities that received AS after 2017 form the control group. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

					1		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	A	nti-immigrati	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Cha	mber of D	eputies		
AS share	0.3775^{***}	0.3209**	0.0526^{**}	-0.0498	-0.2771^{*}	-0.1102^{**}	0.0462
	(0.1294)	(0.1377)	(0.0215)	(0.0777)	(0.1643)	(0.0452)	(0.0369)
long time \times AS share	-0.1799	-0.1876	0.0154	0.0395	0.1025	0.0109	-0.0066
	(0.1448)	(0.1516)	(0.0383)	(0.0887)	(0.1786)	(0.0703)	(0.0537)
				Senate			
AS share	0.4272^{***}	0.2707^{*}	0.1226^{***}	-0.0833	-0.3105**	-0.1022**	0.0386
	(0.1180)	(0.1614)	(0.0415)	(0.0731)	(0.1486)	(0.0503)	(0.0349)
long time \times AS share	-0.2571^{*}	-0.1664	-0.0547	0.1060	0.1270	-0.0406	0.0059
	(0.1351)	(0.1727)	(0.0539)	(0.0950)	(0.1648)	(0.0831)	(0.0527)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	13930	13930	13930	13930	13930	13930	13930

Table A8: Election results - Time exposure

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

Table A9: Election results excluding municipalities with government centers (CARA, CPR, Hotspots)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	nti-immigratio	PDL	M5S	center-left	Turnout	
	Total	Lega Nord	FDI			PD + Left	
			Char	nber of De	eputies		
AS share	0.2396***	0.1716^{***}	0.0686**	-0.0076	-0.2064***	-0.1117**	0.0554
	(0.0561)	(0.0550)	(0.0288)	(0.0416)	(0.0735)	(0.0505)	(0.0356)
				Senate			
AS share	0.2301***	0.1346^{**}	0.0872**	0.0054	-0.2277***	-0.1523**	0.0578
	(0.0591)	(0.0567)	(0.0379)	(0.0560)	(0.0710)	(0.0642)	(0.0362)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13894	13894	13894	13894	13894	13894	13894

	cetton resu	itis CACIUU	ing munic	numerpanetes with large centers (>=100)					
	(1)	(2)	(3)	(4)	(5)	(6)	(7)		
	Ar	nti-immigratio	on	PDL	M5S	center-left	Turnout		
	Total	Lega Nord	FDI			PD + Left			
			Char	nber of De	eputies				
AS share	0.3245^{***}	0.2682^{***}	0.0551^{**}	-0.0138	-0.2807***	-0.0921^{*}	0.0550		
	(0.0665)	(0.0646)	(0.0223)	(0.0428)	(0.0926)	(0.0497)	(0.0387)		
				Senate					
AS share	0.3262^{***}	0.2205***	0.0931**	-0.0385	-0.3170***	-0.1400**	0.0522		
	(0.0604)	(0.0724)	(0.0433)	(0.0440)	(0.0865)	(0.0670)	(0.0380)		
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Ν	13538	13538	13538	13538	13538	13538	13538		

Table A10: Election results – excluding municipalities with large centers (>=100)

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01.AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on the total population, share of foreign born population.

Table A11: Election results - excluding municipalities with a high concentration of asylum seekers (>=99th share percentile)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)		
	А	nti-immigrat	ion	PDL	M5S	center-left	Turnout		
	Total	Lega Nord	FDI			PD + Left			
			Chamber of Deputies						
AS share	0.3006**	0.3753^{***}	-0.0654	-0.1091	-0.2371	-0.2385***	-0.0211		
	(0.1426)	(0.1268)	(0.0525)	(0.0851)	(0.1822)	(0.0865)	(0.0673)		
				Senate					
AS share	0.2334^{*}	0.3388^{***}	-0.1001**	-0.0695	-0.2650	-0.4918^{***}	-0.0345		
	(0.1397)	(0.1237)	(0.0499)	(0.0904)	(0.1832)	(0.1003)	(0.0694)		
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Ν	13792	13792	13792	13792	13792	13792	13792		

					1 1		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	nti-immigrati	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Che	amber of De	eputies		
AS share	0.2460^{***}	0.1836^{***}	0.0639***	-0.0198	-0.2014^{***}	-0.1019^{**}	0.0427
	(0.0509)	(0.0509)	(0.0248)	(0.0366)	(0.0619)	(0.0427)	(0.0315)
top 5% largest municipalities \times AS share	-3.8414^{***}	-3.7032^{***}	0.1430	-1.2159^{**}	0.8775	-0.4461	-2.8176^{***}
	(0.9150)	(0.7316)	(0.2669)	(0.4805)	(1.2138)	(0.4774)	(0.3595)
				Senate			
AS share	0.2382^{***}	0.1490^{***}	0.0818^{**}	-0.0041	-0.2167***	-0.1317^{**}	0.0444
	(0.0526)	(0.0522)	(0.0328)	(0.0480)	(0.0601)	(0.0536)	(0.0320)
top 5% largest municipalities \times AS share	-3.5018^{***}	-3.7312^{***}	0.4022^{*}	-1.1711^{**}	0.9834	-1.1955	-2.7945^{***}
	(0.8848)	(0.7355)	(0.2308)	(0.4881)	(1.1894)	(0.9816)	(0.3605)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table A12: Election results - Interaction with top 5% population size

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	nti-immigratio	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
				Chamber o	f Deputies		
AS share	0.2121^{***}	0.1526^{***}	0.0619^{**}	-0.0142	-0.1709^{***}	-0.1064^{**}	0.0462
	(0.0517)	(0.0508)	(0.0250)	(0.0364)	(0.0600)	(0.0433)	(0.0319)
top 5% share of foreigners \times AS share	2.3489^{***}	2.1432^{***}	0.1569	-0.4594^{*}	-2.2178^{***}	0.3154	-0.3633*
	(0.3553)	(0.3461)	(0.1409)	(0.2445)	(0.4309)	(0.2226)	(0.2172)
				Sen	ate		
AS share	0.2088***	0.1221**	0.0797^{**}	-0.0007	-0.1870***	-0.1355**	0.0475
	(0.0549)	(0.0512)	(0.0334)	(0.0482)	(0.0590)	(0.0544)	(0.0325)
top 5% share of foreigners \times AS share	2.0312^{***}	1.8458^{***}	0.1720	-0.2936	-2.1556^{***}	0.2364	-0.3326
	(0.3731)	(0.3539)	(0.1160)	(0.3089)	(0.4317)	(0.2622)	(0.2402)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table A13: Election results - Interaction with top 5% foreign population share

				-		,,,,,,,	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	An	ti-immigratic	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			C	Chamber of	Deputies		
AS share	0.2609^{***}	0.1963^{***}	0.0652^{**}	-0.0314	-0.1735^{***}	-0.1134^{***}	0.0532^{*}
	(0.0520)	(0.0523)	(0.0254)	(0.0384)	(0.0604)	(0.0437)	(0.0322)
top 5% share of uni degree \times AS share	-0.5288^{***}	-0.4578^{***}	-0.0368	0.3413^{***}	-0.8570^{***}	0.3513	-0.3710^{**}
	(0.1510)	(0.1719)	(0.0626)	(0.1139)	(0.2475)	(0.2271)	(0.1810)
				Senat	e		
AS share	0.2537^{***}	0.1604^{***}	0.0838^{**}	-0.0139	-0.1899***	-0.1409**	0.0527
	(0.0526)	(0.0537)	(0.0334)	(0.0495)	(0.0593)	(0.0553)	(0.0327)
top 5% share of uni degree \times AS share	-0.5419^{***}	-0.4131^{*}	-0.0565	0.2871^{***}	-0.8227^{***}	0.2671	-0.3034^{*}
	(0.1869)	(0.2270)	(0.0521)	(0.1069)	(0.2402)	(0.2499)	(0.1830)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table A14: Election results - Interaction with top 5% university education

Notes: Standard errors in parentheses clustered at the municipality level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at the municipality level. Municipality controls: Log(Income per capita), share of residents over 65 on total population.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Ar	nti-immigratio	on	PDL	M5S	center-left	Turnout
	Total	Lega Nord	FDI			PD + Left	
			Ch	namber of 1	Deputies		
AS share	0.3173^{***}	0.2488^{***}	0.0680***	-0.0195	-0.3073***	-0.0785^{*}	0.0337
	(0.0517)	(0.0533)	(0.0257)	(0.0380)	(0.0748)	(0.0439)	(0.0324)
top 5% unemployment rate \times share AS	-1.4215^{***}	-1.3017^{***}	-0.0772	-0.0182	2.0609^{**}	-0.4589^{*}	0.1478
	(0.4343)	(0.4026)	(0.0707)	(0.1354)	(0.9280)	(0.2492)	(0.1121)
				Senat	е		
AS share	0.3044^{***}	0.2068***	0.0877***	0.0052	-0.3269***	-0.1121**	0.0340
	(0.0497)	(0.0569)	(0.0335)	(0.0498)	(0.0708)	(0.0539)	(0.0329)
top 5% unemployment rate \times share AS	-1.3180^{***}	-1.1575^{***}	-0.1099^{*}	-0.1910	2.1460^{**}	-0.3927	0.1739
	(0.4020)	(0.3717)	(0.0662)	(0.1973)	(0.8877)	(0.3451)	(0.1211)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13930	13930	13930	13930	13930	13930	13930

Table A15: Election results - Interaction with top 5% unemployment rate

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	log(income p.c.)	$\log(\text{income})$	$\log(\text{pop.})$	log(native pop.)	$\log(\text{foreign pop.})$	$\log(\text{native pop.})$	$\log(\text{income})$
		a) Top 5 % a	sylum seekers sha	re municipalities ex	cluded	
AS share	-0.0042**	0.0083***	0.0125***	0.0061***	0.1102***	0.0048***	0.0052***
	(0.0018)	(0.0019)	(0.0015)	(0.0015)	(0.0122)	(0.0011)	(0.0018)
death rate (2013-2017) \times Time						-0.0094***	-0.0109***
						(0.0006)	(0.0007)
birth rate (2013-2017) \times Time						0.0185***	
	37	37	V	37	37	(0.0007)	37
Municipality FE Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N N	13236	13236	13236	13236	13236	13236	13236
	13230	15250	10200	13230	13230	15250	15250
AC -h	0.0025***	0.0010**	b) Go	vernment reception	1 centeres excluded	0.0000	0.0001
A5 snare	-0.0055	-0.0010^{-0}	$(0.0025^{\circ\circ\circ})$	-0.0010	(0.0024)	0.0000	-0.0001
	(0.0009)	(0.0005)	(0.0008)	(0.0005)	(0.0108)	(0.0004)	(0.0000)
death rate (2013-2017) \times Time						-0.0092***	-0.0106***
						(0.0006)	(0.0007)
1. 1 (2012-2017) T.						0.0109***	
birth rate $(2013-2017) \times 11$ me						(0.006)	
Municipality FF	Voc	Voc	Voc	Voc	Voc	(0.0000) Voc	Voc
Time FE	Ves	Ves	Ves	Ves	Ves	Ves	Ves
N	13894	13894	13894	13894	13894	13894	13894
				antana ahava 100	muata avaludad		
AS share	-0.0028***	-0.0009*	0.0019**	-0.0020***	0.0562***	0.0001	0.0003
All share	(0.0008)	(0.0005)	(0.0013)	(0.0007)	(0.0180)	(0.0004)	(0.0007)
	(0.0000)	(010000)	(0.0000)	(0.000)	(0.0200)	(01000-)	(0.000.)
death rate (2013-2017) \times Time						-0.0092***	-0.0108***
						(0.0006)	(0.0007)
birth rate (2013-2017) \times Time						0.0183***	
$\operatorname{Siten}\operatorname{rate}(2019-2011)\times\operatorname{rime}$						(0.0006)	
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13538	13538	13538	13538	13538	13538	13538
				d) labor Market A	rea estimates		
AS share	-0.0080**	-0.0069**	0.0011	-0.0037	0.0693***	0.0002	-0.0036
	(0.0034)	(0.0029)	(0.0027)	(0.0025)	(0.0178)	(0.0020)	(0.0039)
deeth ante (2012-2017) y Time						0.0151***	0.0197***
death rate (2013-2017) \times 11me						-0.0151	-0.0157 (0.0016)
						(0.0010)	(0.0010)
birth rate (2013-2017) \times Time						0.0049^{***}	
						(0.0017)	
LMA FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	1122	1122	1122	1122	1122	1122	1122

Table A16: Effect on income and population - Robustness checks at LMA level

Notes: Standard errors in parentheses clustered at municipality/LMA level. * p < 0.10, ** p < 0.05, *** p < 0.01. AS share is the fraction of asylum seekers over total baseline population at municipality/LMA. Death rate (2013-2017) is the municipality/LMA fraction of Italian citizens who died between 2013 and 2017 over the 2013 population. Birth rate (2013-2017) is the municipality/LMA fraction of Italian citizens born between 2013 and 2017 over the 2013 population. Both are interacted with a year dummy in order to compare municipalities/LMAs having similar levels of the two variables in the treatment period (2013-2017).

			1 0	· 1			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Log(Total)	Log(Families)	Log(Addicted)	Log(Poverty)	Log(Elderly)	Log(Disabilities)	Log(Immigrants)
AS share	-0.0014	-0.0052	0.0057^{***}	0.0074	0.0009	-0.0080	0.0085
	(0.0068)	(0.0075)	(0.0022)	(0.0079)	(0.0104)	(0.0316)	(0.0108)
Municipality controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipality FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	13604	13604	13604	13604	13604	13604	13604

Table A17: Municipality expenditure - controls